

Moments in Pearson's Four-Step Uniform Random Walk Problem and Other Applications of Very Well-Poised Generalized Hypergeometric Series

J. Roderick McCrorie University of St Andrews, Scotland, UK

Abstract

This paper considers the representation of odd moments of the distribution of a four-step uniform random walk in even dimensions, which are based on both linear combinations of two constants representable as contiguous very well-poised generalized hypergeometric series and as even moments of the square of the complete elliptic integral of the first kind. Neither constants are currently available in closed form. New symmetries are found in the critical values of the L-series of two underlying cusp forms, providing a sense in which one of the constants has a formal counterpart. The significant roles this constant and its counterpart play in multidisciplinary contexts is described. The results unblock the problem of representing them in terms of lower-order generalized hypergeometric series, offering progress towards identifying their closed forms. The same approach facilitates a canonical characterization of the hypergeometry of the parbelos, adding to the characterizations outlined by Campbell, D'Aurozio and Sondow (2020, The American Mathematical Monthly 127(1), 23-32). The paper also connects the econometric problem of characterizing the bias in the canonical autoregressive model under the unit root hypothesis to very well-poised generalized hypergeometric series. The confluence of ideas presented reflects a multidisciplinarity that accords with the approach and philosophy of Prasanta Chandra Mahalanobis.

Keywords. Four-step uniform random walk in the plane, Dickey-Fuller distribution, very well-poised generalized hypergeometric series, elliptic integral, universal parabolic constant, moments AMS (2000) subject classification. Primary 33C20, 60G50, 62M10, Secondary 11Y60

1 Introduction

In his assessment of the impact of Karl Pearson's work in the development of Statistics in India, Nayak (2009) outlined the nascent role it played in attracting Prasanta Chandra Mahalanobis to the discipline, and in inspiring

him towards his eventual leading role in its development across the Indian subcontinent. See also Ghosh (1994). C.R. Rao (1973) saw Mahalanobis existentially as ". . . a physicist by training, a statistician by instinct and a planner by conviction." He argued that

"He did not consider statistics as a narrow subject confined to the mathematical theory of probability, or routine analysis of data in applied research, or collection of data as an aid to administrative decisions . . . But he took a wider view of statistics as 'a new technology for increasing the efficiency of human efforts in the widest sense'. This has naturally aroused his interest in various fields and enabled him to enrich the science of statistics with a practical base of great depth and spread. (*ibid.*, p. 463)

This paper is also influenced by an early Pearson contribution: a statement of the problem on the random walk. While it does not deal directly with an issue inspired by a statistical contribution by Mahalanobis and has more scientific than practical application, it attempts to offer a multidisciplinary approach, including an application to Econometrics, that is consistent with his wider statistical philosophy.

Pearson (1905) initiated work on a now-celebrated problem: characterizing the distribution of the distance travelled in an N-step random walk in the plane. The walk starts at the origin and entails N steps of unit length, each taken in a uniformly random direction. The problem is classical and attracted some immediate contributions, e.g. Rayleigh (1905) who, referring to his earlier work, gave the distribution for large N; and Kluvver (1905) who expressed the density function for general N in terms of an integral involving Bessel functions. Finding closed-form expressions for the densities for small N, however, has proved to be considerably more challenging, given their lack of smoothness and the radical differences in their shapes compared with the densities for larger N. Indeed, it could be argued the small-N problem has only received a satisfactory treatment given recent developments in mathematical and computational technology. General approaches to the small-N problem are Borwein, Nuyens, Straub and Wan (2011), Borwein, Straub, Wan and Zudilin (2012), Borwein, Straub and Vignat (2016) and Joyce (2017). Related papers, including some of a more technical nature, are Borwein, Straub and Wan (2013), Borwein and Straub (2013), Borwein and Sinnamon (2016) and Zhou (2019a). Borwein (2016) offers an introduction and perspective on recent work. The problem is also discussed in Chapter 6 of the recent monograph by Brunault and Zudilin (2020).

We first consider the N-step uniform random walk problem in the plane, although our results pertain to higher dimensions. Let X_N be the distance to the origin after N steps. Borwein, Straub, Wan and Zudilin (2012) showed the s-th moments $\mathscr{M}_N(s)$ of X_N can be computed by

$$\mathscr{M}_{N}(s) = \int \dots \int_{[0,1]^{N}} \left| e^{2\pi i \theta_{1}} + \dots + e^{2\pi i \theta_{N}} \right|^{s} d\theta_{1} \dots d\theta_{N}.$$
(1.1)

Let $p_N(x)$ be the probability density function of X_N . Then $p_N(x)$ and $\mathcal{W}_N(s)$ are related via

$$\mathscr{W}_{N}(s) = \int_{0}^{\infty} x^{s} p_{N}(x) dx = \int_{0}^{N} x^{s} p_{N}(x) dx, \qquad (1.2)$$

where, accordingly, $\mathcal{W}_{N}(s-1)$ is the Mellin transform of $p_{N}(x)$. Kluyver (1905) showed that

$$p_N(x) = \int_0^\infty xt J_0(xt) J_0(t)^N dt, \qquad (1.3)$$

where $J_0(t) \coloneqq \frac{2}{\pi} \int_0^{\pi/2} \cos(t \cos \theta) \, d\theta$ is the Bessel function of the first kind and zero-th order.

As a probability density function, $p_N(x)$ is supported on [0, N]. While closed forms are available for N = 1 and 2, the small-N behaviour of (1.3) makes finding closedforms or even just tractable expressions for the densities and moments in the shortwalk problems analytically and numerically challenging. This is especially so for N = 3, 4, 5 and 6. Borwein, Nuyens, Straub and Wan (2011) illustrate the dramatically different shapes of the small-N density functions, including the so-called "shark-fin curve" of the N = 4 case of interest here. The densities become smooth only at N = 6. Remarkably, the small-N case has thrown up features that have numerous analogies in multidisciplinary contexts, including Analytic Number Theory and Physics.

Borwein, Nuyens, Straub and Wan (2011) showed by two different methods that for general N in the plane,

$$\mathcal{M}_{N}(s) = N^{s} \sum_{m=0}^{\infty} (-1)^{m} {\binom{s/2}{m}} \sum_{k=0}^{m} \frac{(-1)^{k}}{N^{2k}} {\binom{m}{k}} \sum_{a_{1}+\ldots+a_{N}=k} {\binom{k}{a_{1},\ldots,a_{N}}}^{2}, \operatorname{Re}(s) > 0,$$
(1.4)

where $\binom{k}{a_1,...,a_N}$ is the multinomial coefficient $\frac{k!}{a_1!...a_N!}$ and the function Re(.)

denotes the real part of a complex number. The even moments take on a simpler form. Borwein, Straub, Wan and Zudilin (2012) showed that

$$\mathscr{M}_{N}(2s) = \sum_{a_{1}+\ldots+a_{N}=s} \begin{pmatrix} s \\ a_{1}, \ldots, a_{N} \end{pmatrix}^{2}.$$
 (1.5)

Borwein, Nuyens, Straub and Wan (2011) found this same expression in Richmond and Shallit (2009), indicating that the even moments in the plane count Abelian squares. They also satisfy a recursion that has an analytic continuation to the complex plane. In the N = 4 case,

$$(s+4)^{3} \mathscr{W}_{4}(s+4) - 4(s+3)(5s^{2}+30s+48) \mathscr{W}_{4}(s+2) + 64(s+2)^{3} \mathscr{W}_{4}(s) = 0.$$
(1.6)

The analytic continuation, based on a three-term recurrence relation, implies that the odd moments can be written in terms of two initial values:

$$\mathscr{W}_{4}(-1) = \int_{0}^{\infty} J_{0}(t)^{4} dt = \frac{4}{\pi^{3}} \int_{0}^{\infty} I_{0}(t) K_{0}(t)^{3} dt, \qquad (1.7)$$

and

$$\mathscr{W}_{4}(1) = 4 \int_{0}^{\infty} J_{1}(t) J_{0}(t)^{3} \frac{dt}{t} = \frac{256}{\pi^{3}} \int_{0}^{\infty} t^{2} I_{0}(t) K_{0}(t)^{3} dt, \qquad (1.8)$$

where $J_0(t)$ are the Bessel functions of the first kind in (1.3), $J_1(t) \coloneqq -dJ_0(t)/dt$, and $I_0(t) \coloneqq \frac{1}{\pi} \int_0^{\pi} \exp(t\cos\phi) \ d\phi$ and $K_0(t) \coloneqq \int_0^{\infty} \exp(-t\cosh u) \ du$ are the modified Bessel functions of first and second kinds, of zero-th order respectively.¹ Borwein, Straub and Vignat (2016) showed through a generalization of (1.5) and (1.6) that the constants given by (1.7) and (1.8) provide a basis for all the odd moments in arbitrary even dimensions in the sense that all such moments are linear combinations of them with weights that are rational numbers.

The primary purpose of the paper is to develop the connection between moments in the four-step random walk problem and very well-poised (VWP) generalized hypergeometric series. The search for closed forms for (1.7) and (1.8) has been an important element of this problem because of the connection it provides to numerous other disciplines. Here, we contribute directly to the problem in two ways. Firstly, we provide a sense in which the constant represented by (1.7) has a formal counterpart in generalized hypergeometric series and describe the multidisciplinary role that both play; and secondly, using methods based on elliptic integrals, we derive new series representations for the constant and its counterpart

¹ The expressions involving I_0 and K_0 are included to elicit a connection to work by Bailey, Borwein, Broadhurst and Glasser (2008) on Feynman diagrams in two-dimensional Quantum Field Theory; Laporta (2008, 2018) on four-loop integrals in Quantum Electrodynamics; and Broadhurst (2013, 2016), Broadhurst and Mellit (2016), Broadhurst and Roberts (2018) and Zhou (2019b) on critical *L*-values attached to modular forms (as defined in Section 2).

that provide another step towards finding their closed forms. The efficacy of the method proposed in the paper is demonstrated through its successful application to the problem of characterizing the hypergeometry of the parbelos recently discussed by Campbell, D'Aurizio and Sondow (2020). Finally, a connection is made between the econometric problem of characterizing the bias in the canonical autoregressive model under the unit root hypothesis and VWP generalized hypergeometric series. This elucidates a shared characteristic with the four-step random walk problem that could be exploited in future work. While the methods presented here are appropriate to the times, they represent a confluence of ideas having a flavour of multidisciplinarity in the spirit of the work and philosophy of Prasanta Chandra Mahalanobis.

The paper is organized as follows. Section 2 provides the notation and definitions of generalized hypergeometric series and the other machinery, including modular forms, used in the later sections. Section 3 contains the main results, providing new expressions for the moment (1.7) and its counterpart defined there. Section 4 contains the application to the hypergeometry of the parbelos. Section 5 offers the econometric application. Section 6 concludes and offers suggestions for further work. Proofs are given in the Appendix.

2 Notation

As usual, we let $N := \{1, 2, ...\}$ be the set of natural numbers, Z the set of integers, Q the set of rational numbers, R the set of real numbers and C the set of complex numbers.

2.1 Generalized Hypergeometric Series and Special Functions For $p, q \in \mathbb{N}_0 := \mathbb{N} \cup \{0\}$, and interpreting empty products as 1,

$$pFq(\alpha_1, \dots, \alpha_p; \beta_1, \dots, \beta_p; z) \equiv pFq\begin{bmatrix}\alpha_1, \dots, \alpha_p\\\beta_1, \dots, \beta_q \end{bmatrix} \coloneqq \sum_{n=0}^{\infty} \frac{(\alpha_1)_n \dots (\alpha_p)_n}{(\beta_1)_n \dots (\beta_q)_n} \frac{z^n}{n!}$$
(2.1)

 $(p \leq q+1; p \leq q \ and \ |z| < \infty; p = q+1 \ and \ |z| < 1; p = q+1; |z| = 1 \ and \ \operatorname{Re}(s) > 0),$

where Pochhammer's symbol $(\lambda)_n$ denotes the shifted factorial function

$$(\lambda)_n = \begin{cases} 1 & (n = 0, \lambda \in \mathbb{C} \setminus \{0\}) \\ \lambda(\lambda + 1) \dots (\lambda + n - 1) & (n \in \mathbb{N}, \lambda \in \mathbb{C}) \end{cases},$$
(2.2)

which, in terms of the gamma function, $\Gamma(z) \coloneqq \int_0^\infty e^{-t} t^{z-1} dt$ (Re(z) > 0), is given by

$$(\lambda)_n = \frac{\Gamma(\lambda + n)}{\Gamma(\lambda)} \qquad (\lambda \in \mathbb{C} \setminus \mathbb{Z}_0^-, \mathbb{Z}_0^- \coloneqq \{0, -1, -2, \ldots\}), \qquad (2.3)$$

This allows us to define the binomial coefficient as

$$\binom{\lambda}{n} \coloneqq \frac{\Gamma(\lambda+1)}{n!\Gamma(\lambda-n-1)} = \frac{(-1)^n (-\lambda)_n}{n!}.$$
(2.4)

The parameter excess, ω , is

$$\omega = \sum_{j=1}^{q} \beta_j - \sum_{j=1}^{p} a_j \left(\beta_j \notin Z_0^-; j = 1, ..., q \right)$$
(2.5)

Specifically, when p = q + 1, (2.1) is

- (i) absolutely convergent for |z| = 1 if $\operatorname{Re}(\omega) > 0$;
- (ii) conditionally convergent for $|z| = 1(z \neq 1)$ if $-1 < \text{Re}(\omega) \le 0$; and

(iii) divergent for |z| = 1 if $\operatorname{Re}(\omega) \leq -1$.

Both standard notations in (2.1) will be used: the second emphasizes the distinction between numerator and denominator parameters but the first is more parsimonious.

We now consider special cases where the numerator and denominator parameters are related in certain ways. The generalized hypergeometric series (2.1), with p = q + 1,

$${}_{q+1}F_q\left[\begin{array}{c} \alpha_1,...,\alpha_{q+1}\\ \beta_1,...,\beta_q \end{array};z\right] = \sum_{n=0}^{\infty} \frac{(\alpha_1)...(\alpha_{q+1})}{(\beta_1)...(\beta_q)} \frac{z^n}{n!},$$
(2.6)

is said to be

(i) balanced if $\alpha_1 + \ldots + \alpha_{q+1} + 1 = \beta_1 + \ldots + \beta_q$;

(ii) nearly-poised (of the first kind) if $\alpha_1 + \beta_1 = ... = \alpha_q + \beta_q$;

(iii) well-poised if $\alpha_{q+1} + 1 = \alpha_1 + \beta_1 = \ldots = \alpha_q + \beta_q$; and

(iv) very well-poised (VWP) if it is well-poised and $\alpha_1 = \frac{1}{2}\alpha_{q+1} + 1$.

Our focus here is on VWP non-terminating $_7F_6(1)$ series,² which take the general form

² The literature on generalized hypergeometric series divides between non-terminating (or never-ending) series and terminating series where the series terminates after a finite number of terms. The assumption in the terminating case that one of the numerator entries c, d, e, for g is a negative integer is therefore not made here.

MOMENTS IN PEARSON'S FOUR-STEP...

$$W(a; c, d, e, f, g) \coloneqq_7 F_6 \begin{bmatrix} a, 1 + \frac{1}{2}a, c, d, e, f, g \\ \frac{1}{2}a, 1 + a - c, 1 + a - d, 1 + a - e, 1 + a - f, 1 + a - g \\ \end{bmatrix}, (2.7)$$

with s = c + d + e + f + g - 2a - 1 being subject to the convergence condition s < 1.

Bailey (1935) showed in his classic tract that, under conditions, a series such as (2.7) can be decomposed into two balanced ${}_{4}F_{3}(1)$ series. See the Appendix for details.

Other special functions of interest include the complete elliptic integral of the first and second kinds:

$$K(k) = \int_0^{\pi/2} \frac{d\theta}{\sqrt{1 - k^2 \sin^2 \theta}} = \frac{1}{2} \pi_{-2} F_1 \left[\frac{1}{2}, \frac{1}{2}; k^2 \right], \qquad |k| < 1, \qquad (2.8)$$

and

$$E(k) = \int_0^{\pi/2} \sqrt{1 - k^2 \sin^2 \theta} \, d\theta = \frac{1}{2} \pi_{-2} F_1 \begin{bmatrix} -\frac{1}{2}, \frac{1}{2}; k^2 \\ 1 \end{bmatrix}, \quad |k| < 1, \quad (2.9)$$

where the real number k is called the *modulus* of the elliptic integral. $K'(k) \coloneqq K(k')$ and $E'(k) \coloneqq E(k')$ are complementary functions, with $k' \coloneqq \sqrt{1-k^2}$ called the complementary modulus.

2.2 Modular forms and L-series Let $SL_2(\mathbb{Z})$ denote the special linear group

$$SL_2(\mathbb{Z}) = \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} : a, b, c, d \in \mathbb{Z} \text{ and } ad - bc = 1 \right\}.$$
 (2.10)

For $k \in \mathbb{Z}$, a modular form of weight k is an analytic function f defined on the upper half plane $H = \{z \in \mathbb{C} : \text{Im}(z) > 0\}$ that transforms according to the rule

$$f\left(\frac{az+b}{cz+d}\right) = (cz+d)^k f(z) \quad \text{for all} \quad \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{Z}), \tag{2.11}$$

and whose Fourier series

$$f(z) = \sum_{n=-\infty}^{\infty} \gamma(n) \exp(2\pi i z), \qquad i = \sqrt{-1}, \tag{2.12}$$

satisfies $\gamma(n) = 0$ for all n < 0. If, in addition, $\gamma(0) = 0$, then f is said to be a cusp form of weight k. For $\text{Im}(\tau) > 0$, the Dedekind eta function is defined by

$$\eta(\tau) \coloneqq q^{1/24} \prod_{m=1}^{\infty} (1 - q^m) = \sum_{n=-\infty}^{\infty} (-1)^n q^{(6n+1)^2/24}, \text{ where } q = \exp(2\pi i \tau).$$
(2.13)

This function offers a way of generating modular forms. See Martin (1996) for a classification.

Given $N \in \mathbb{N}$, we can define the following congruence subgroup

$$\Gamma_0(N) \coloneqq \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{Z}) : c \equiv 0 \mod N \right\}.$$
 (2.14)

Then a weight k modular form (respectively, cusp form) of level N, $f_{k, N}$, is defined as before, but with $SL_2(\mathbb{Z})$ replaced by $\Gamma_0(N)$. Its *L*-function can be defined through a Mellin transform:

$$L(f_{k,N},s) \coloneqq \frac{(2\pi)^s}{\Gamma(s)} \int_0^\infty f_{k,N}(iy) y^{s-1} dy.$$
(2.15)

A special *L*-value $L(f_{k, N}, s_0)$ is said to be *critical* if $s_0 \in \mathbb{N} \cap [1, k-1]$. See Koblitz (1993) for more background to this section.

2.3 Basis constants and integer-relation algorithms Our generic problem is whether the generalized hypergeometric series involved in the representation of the moments of the four-step random walk problem via (1.7) and (1.8) can be evaluated in terms of known constants or reduced to a more fundamental, tractable or accessible form. Moments are, of course, just numbers and so notions of a "closed form" of a number, as discussed by Borwein and Crandall (2013), are important. Here, we shall work within what they called the *ring of hyperclosure*, which essentially means numbers or constants that are representable as generalized hypergeometric series. An important question then is whether such numbers are *further reducible* as products of gamma functions of rational argument, or more generally in terms of a linear combination of a set of notionally fundamental constants, including such gamma values, with weights in Q. The general question of identifying such fundamental constants, or *basis constants*, has become an important topic in Quantum Field Theory. See, e.g., Ablinger and Blümlein (2013), Ablinger, Blümlein and Schneider (2011, 2013) and Laporta (2018). The premise that basis constants exist is entirely consistent with the philosophy of integerrelation algorithms such as the PSLQ algorithm (Bailey and Broadhurst, 2000). Given a set of real numbers known to a given precision, such

algorithms involve a search that seeks integer relations among them, or seeks to determine that no such relation exists within a certain bound.³ McCrorie (2020b) takes the idea a step farther and explicitly proposes a notion of *duality* between series and integrals on one hand and basis constants on the other. Instead of just classically summing series and evaluating integrals in terms of rational linear combinations of basis constants, the dual approach sees linear combinations of basis constants *decomposed* into series and integrals in such a way that different series and integrals can be systematically and taxonomically related. The potential relevance to the current problem is demonstrated in Section 4 where it is successfully applied to construct the hypergeometry of the parbelos, albeit in a context where closed form summation is already available.

The following constants, which are *periods* in the sense of Kontsevich and Zagier (2001), are treated here as basis constants:

$$\sqrt{2} := \int_{x^2 \le \frac{1}{2}} dx \cong 1.41421 \ 35623 \ 73095 \ 04880 \ 16887 \ 24209 \ 69807 \ 85696... \tag{2.16}$$

$$\pi \coloneqq \iint_{x^2 + y^2 \le 1} dx \ dy = \Gamma\left(\frac{1}{2}\right)^2 \cong 3.14159\ 26535\ 89793\ 23846\ 26433\ 83279\ 502... \tag{2.17}$$

$$L = \iint_{x^4 + y^4 \le 1} dx \, dy = \Gamma\left(\frac{1}{4}\right)^2 / 2\sqrt{2\pi} \approx 2.62205\ 75542\ 92119\ 81046\ 48395\ 89... \tag{2.18}$$

$$\boldsymbol{\varpi}_{8} \coloneqq \iint_{x^{8}+y^{8} \le 1} dx \ dy = \Gamma\left(\frac{1}{8}\right)^{2} / 2^{7/4} \Gamma\left(\frac{1}{4}\right) \cong 4.65437\ 02848\ 73077\ 50121\ 007257... \quad (2.19)$$

$$\log 2 \coloneqq \int_{0 < x < \frac{1}{2}} \frac{dx}{1 - x} \cong 0.69314\ 71805\ 59945\ 30941\ 72321\ 21458\ 17656\ 80755... \tag{2.20}$$

$$\log\left(1+\sqrt{2}\right) \coloneqq \int_{0 < x < 1} \frac{dx}{\sqrt{1+x^2}} \approx 0.88137\ 35870\ 19543\ 02523\ 26093\ 24979\ 79230... \tag{2.21}$$

The constants π , L and $\overline{\omega}_8$ have the geometric interpretation of being arclengths of the circle, lemniscate and quatrefoil respectively. The algebraic independence of π and L follows from Chudunovsky's theorem which established the algebraic independence of $\Gamma(\frac{1}{2})$ and $\Gamma(\frac{1}{4})$ (see Waldschmidt, 2008). In accordance with the approach taken in integer-relation algorithms, we shall avoid using mixed expressions involving, say, π and $\Gamma(\frac{1}{4})$, or using $\Gamma(\frac{1}{4})$ alongside $\Gamma(\frac{3}{4})$, and instead use π , Land $\overline{\omega}_8$, and their powers and reciprocal powers, as basis constants.⁴ McCrorie

³ While any positive result represents, at best, the basis of a conjecture, through the identification process such an algorithm may *facilitate* the construction of a formal proposition and a proof or refutation.

⁴ For example, instead of $\Gamma(\frac{3}{4})$, we use $\pi\sqrt{2}/\Gamma(\frac{1}{4})$, through which a term in the *reciprocal* of L emerges.

(2020b) discusses a possible alternative approach that utilizes the singular values of the elliptic integral K. Both approaches are essentially based on values of the central beta function and, while the singular value approach would in certain circumstances usefully facilitate a connection with theta functions, they are not required here.

3 Main results

In this section, we first state an expression for the density p_4 and the series governing its odd moments in even dimensions. We then show how the series for $\mathscr{W}_4(-1)$ has a counterpart in analysis that arises through the consideration of an intertwinement between certain cusp forms. New symmetries between the pair of constants are then established, which at the same time reveal new expressions for each in terms of fundamental VWP₇ $F_6(1)$ series that are reducible to sums of lower-order ${}_4F_3(1)$ series with entries in $\frac{1}{4}$ N and $\frac{1}{8}$ N.

3.1 Generalized hypergeometric series representation of even moments in the four-step random walk problem With the notation in Section 2, we begin with the expression for p_4 derived by Borwein, Straub, Wan and Zudilin (2012), which is valid for all $x \in (0, 4]$:

$$p_4(x) = \left(\frac{2}{\pi^2}\right) \frac{\sqrt{(4-x)(4+x)}}{x} \operatorname{Re}_3 F_2 \left[\frac{\frac{1}{2}}{\frac{1}{2}}, \frac{1}{2}, \frac{1}$$

The above expression embodies an analytic continuation as the ${}_{3}F_{2}$ function on the interval (0, 2) is complex, and masks the fact that the expansion of p_{4} at 0 involves logarithmic terms. Joyce (2017) discusses this aspect in some detail and provides other expressions for p_{4} . Unfortunately, none of the hitherto derived expressions for p_{4} readily lends itself towards deriving tractable expressions for the moments of p_{4} via a direct application of (1.2). The even moments in odd dimensions can be expressed via (1.7) and (1.8) in terms of linear combinations VWP nonterminating ${}_{7}F_{6}(1)$ series, viz. Moments in Pearson's Four-Step...

11

$$\mathcal{W}_{4}(-1) = \frac{\pi}{4} W\left(\frac{1}{2}; \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\right)$$

$$= \frac{\pi}{4} {}_{7}F_{6}\left[\begin{array}{cccc} \frac{1}{2}, \frac{5}{4}, \frac{1}{2}, \frac{1$$

and

$$\mathscr{W}_{4}(1) = 4\mathscr{W}_{4}(-1) - \frac{9\pi}{16}W\left(\frac{3}{2};\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{3}{2}\right)$$
(3.3)

See Borwein, Straub and Vignat (2016) for details. To focus on the topics of interest in the following section, we shall only concentrate on the series in (3.2). Further results, including a discussion of (3.3), are provided in the companion paper by McCrorie (2020a) where series allied to the series in (3.2) and (3.3) are constructed taxonomically.

3.2 $\mathscr{W}_4(-1)$ and a counterpart as critical values of the L-series of cusp forms Consider the pair of weight 4 cusp forms

$$f_{4,8}(\tau) = \eta(2\tau)^4 \eta(4\tau)^4, \quad f_{4,16}(\tau) = \frac{\eta(4\tau)^{16}}{\eta(2\tau)^4 \eta(8\tau)^4}, \tag{3.4}$$

with levels 8 and 16 (LMFDB labels 8.4.1a and 16.4.1.a) respectively.⁵ Rogers, Wan and Zucker (2015, p.128) noted the following intertwinement among their critical L-values:⁶

$$L(f_{4,16},3) = \frac{\pi}{2} L(f_{4,8},2) = \frac{\pi^2}{8} L(f_{4,16},1) = \frac{1}{4} \int_0^1 K(k) K'(k) \, dk = \frac{1}{8} \int_0^1 \frac{K(k)^2}{k'} \, dk \qquad (3.5)$$
$$= \frac{\pi^3}{32} 4F_3 \left[\frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, 1 \right]$$

and

⁵ The LMFDB collaboration provides the *L*-functions and Modular Forms Database, where modular forms and their properties are classified by label. See: http://www.lmfdb.org.

 $^{^6}$ The penultimate expression of (3.5) has been replaced by an equivalent expression involving K and K' that arguably makes the symmetry clearer.

$$L(f_{4,8},3) = \frac{\pi}{2} L(f_{4,16},2) = \frac{\pi^2}{4} L(f_{4,8},1) = \frac{1}{8} \int_0^1 \frac{K(k)K'(k)}{k'} dk = \frac{1}{4} \int_0^1 K(k)^2 dk$$
$$= \frac{\pi^4}{128} {}_7F_6 \begin{bmatrix} \frac{1}{2}, \frac{5}{4}, \frac{1}{2}, \frac{1}{$$

where K(k) is the complete integral of the first kind and K'(k) is its complementary function. The constants represented by the generalized hypergeometric series in (3.5) and (3.6) appear in a number of different fields. Laporta (2008) treated both constants as simplest cases in his analysis of spin integrals related to Feynman diagrams. His constant A is a multiple $\pi^3/8$ of the $_4F_3(1)$ series in (3.5) and his constant B is a multiple $\pi^4/16$ of the $_7F_6(1)$ series in (3.6). As noted, Rogers, Wan and Zucker (2015) found both constants in the L-series evaluations discussed above, as did Wan and Zucker (2016) in certain eight-dimensional lattice sum evaluations. The first constant is a multiple of the constant $c_{4,0} = \int_0^\infty K_0(t)^4 dt$ that arose in evaluations by Bailey, Borwein, Broadhurst and Glasser (2008) of integrals that arise in Quantum Field Theory. Seen as a four-loop sunrise integral, it relates directly to Laporta's (2008) paper and indeed their constant $c_{4,0}$ is a multiple of 2π of Laporta's constant A. The second constant here, represented by the VWP $_7F_6(1)$ series in (3.6), is a multiple of their constant $s_{4,0}$ which is expressed in terms of the integral in (1.7) involving modified Bessel functions. Their $s_{4,0}$ is the same as Laporta's constant B. Guttmann (2010) demonstrated a connection of the first constant to the lattice Green's function of the four-dimensional hyper-body-centred cubic lattice. It has also appeared recently, and more fundamentally, in the areas of supercongruences and Calabi-Yau threefolds (Zagier, 2018; Zudilin, 2018; and Osburn and Straub, 2019). The second constant, of direct interest here given its role in the four-step random walk problem, is central in the theory of elliptic integrals. See especially Wan (2012).

The possibility of whether the two generalized hypergeometric series can be reduced to Q.-linear spans of sets of more basic mathematical constants remains an open problem and, indeed, the hypergeometric form in (3.2) has not hitherto been found to be especially amenable to decomposition or analysis. Borwein and Straub (2013) note that an immediate expression in terms of balanced ${}_{6}F_{5}(1)$ series is available via standard contiguous series relations:

$$W\left(\frac{1}{2};\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}\right) = {}_{6}F_{5}\left[\frac{1}{2},$$

Following an intensive search based on the PSLQ algorithm, Wan (2013) and Borwein, Straub and Wan (2013) found a conjectural three-term relation whose validity they then established using known integrals associated with W. Zudilin and Y.V. Nesterenko in work on the irrationality of special odd values of the Riemann zeta function and related constants:⁷

$$W\left(\frac{1}{2};\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}\right) = \frac{3}{16} W\left(\frac{3}{2};\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{3}{2}\right) + \frac{3}{4} W\left(\frac{3}{2};\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{3}{2},\frac{3}{2}\right).$$
(3.8)

The first right-hand-side series in involved in (3.3) but, again, neither series on the right-hand-side readily decomposes into two balanced $_4F_3(1)$ series using standard methods.

Our principal result shows that lower-order generalized hypergeometric series are available for both constants and that there is a formal sense in which they are *exact counterparts of each other*. This idea is implicit in certain decompositions of integrals of elliptic integrals in the Ph.D. thesis by Wan (2013) although the idea was not developed there. See also Wan (2012). Expression (3.9) below in its essential form was obtained by Wan (2013, p. 115) as an expression involving ${}_{4}F_{3}(1)$ series, although here we show it is more fundamentally an expression involving a three-term relation between VWP ${}_{7}F_{6}(1)$ series. (3.10) is new. Decomposing each series into two balanced ${}_{4}F_{3}(1)$ series reveals a remarkable symmetry: a chosen multiple of the ${}_{7}F_{6}(1)$ series, the magnitudes of whose weights are *exactly the same*.

THEOREM 1.

(a)
$$\frac{\pi^3}{4} W\left(\frac{1}{2}; \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\right) = L^2 W\left(\frac{1}{4}; \frac{1}{8}, \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{5}{8}\right) + \frac{\pi^2}{L^2} W\left(\frac{3}{4}; \frac{3}{8}, \frac{1}{2}, \frac{1}{2}, \frac{3}{4}, \frac{7}{8}\right)$$
(3.9)

$$\frac{\pi^2}{2} W\left(\frac{1}{2}; \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{3}{4}\right) = L^2 W\left(\frac{1}{4}; \frac{1}{8}, \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{5}{8}\right) - \frac{\pi^2}{L^2} W\left(\frac{3}{4}; \frac{3}{8}, \frac{1}{2}, \frac{1}{2}, \frac{3}{4}, \frac{7}{8}\right)$$
(3.10)

 $^{^{7}}$ See Borwein, Straub and Wan (2013).

(b)
$$\pi^{2} \mathscr{W}_{4}(-1) = \frac{8}{\pi} \int_{0}^{1} K(k)^{2} dk = \frac{32}{\pi} L(f_{4,8},3) = 8L(f_{4,16},2) = 8\pi L(f_{4,8},1)$$
$$= \frac{\pi^{3}}{4} {}_{7}F_{6} \begin{bmatrix} \frac{1}{2}, \frac{5}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2} \\ \frac{1}{4}, 1, 1, 1, 1, 1, 1 \end{bmatrix}$$
$$= L^{2} {}_{4}F_{3} \begin{bmatrix} \frac{1}{4}, \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2} \\ \frac{3}{4}, \frac{3}{4}, 1 \end{bmatrix} + \frac{\pi^{2}}{L^{2}} {}_{4}F_{3} \begin{bmatrix} \frac{1}{2}, \frac{1}{2}, \frac{3}{4}, \frac{3}{4} \\ 1, \frac{5}{4}, \frac{5}{4} \end{bmatrix}$$
(3.11)

$$\frac{16}{\pi}L(f_{4,16},3) = 8L(f_{4,8},2) = 2\pi L(f_{4,16},1) = \frac{\pi^2}{2} {}_4F_3 \left[\frac{1}{2},\frac{1}{$$

Theorem 1 expresses the same idea in two ways. In part (a), a chosen multiple of each VWP series is seen to satisfy a three-term relation involving the same VWP $_7F_6(1)$ series, whose weights, L^2 and π^2/L^2 , are the same up to a sign. The series $W(\frac{1}{4}; \frac{1}{8}, \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{5}{8})$ and $W(\frac{3}{4}; \frac{3}{8}, \frac{1}{2}, \frac{1}{2}, \frac{3}{4}, \frac{7}{8})$ relate to each other in the same way that π relates to $1/\pi$ and $\Gamma(\frac{1}{4})$ relates to $\Gamma(\frac{3}{4})$. Note that towards the ultimate aim of obtaining a closed form for both left-hand-side series, the decomposition on the right-hand-side has entries in $\frac{1}{8}$ N. Part (b) expresses the same idea in $_4F_3(1)$ series. While the above results unblock the problem of finding decompositions of $W(\frac{1}{2}; \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2})$ in terms of entries in $\frac{1}{4}$ N and $\frac{1}{8}$ N, none of the series on the right-hand-side of the above expressions has a known closed form.

A further decomposition of the series in (3.5) and (3.6) can be obtained using the decomposition of the moments of a different elliptic integral.

THEOREM 2.

(a)
$$\frac{\pi^4}{2} W\left(\frac{1}{2}; \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\right) = L^4 W\left(\frac{1}{4}; \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\right) - 3\left(\frac{\pi}{L}\right)^4 W\left(\frac{3}{4}; \frac{3}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\right)$$
(3.13)

$$2\pi^{3}W\left(\frac{1}{2};\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{3}{4}\right) = L^{4}W\left(\frac{1}{4};\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}\right) + 3\left(\frac{\pi}{L}\right)^{4}W\left(\frac{3}{4};\frac{3}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}\right) \quad (3.14)$$

(b)
$$\pi^3 \mathcal{W}_4(-1) = 8 \int_0^1 K(k)^2 dk = 32L(f_{4,8},3) = 8\pi L(f_{4,16},2) = 8\pi^2 L(f_{4,8},1)$$

$$=2L_{4}^{3}F_{3}\left[\frac{1}{4},\frac{1}{4},\frac{1}{4},\frac{1}{4},\frac{1}{4};1\right]+\left(\frac{\pi}{L}\right)^{3}F_{3}\left[\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4};1\right]\\-\pi^{2}F_{4}F_{3}\left[\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2};1\right].$$

$$32L(f_{4,16},3) = 16\pi L(f_{4,8},2) = 4\pi^2 L(f_{4,16},1) = \pi^3 {}_4F_3 \left[\frac{1}{2}, \frac{1}{2},$$

Again, the decomposition is seen fundamentally to involve three-term relations involving VWP $_7F_6(1)$ series and involves remarkable symmetries. While there is a generalized hypergeometric series in (3.15) that appears not to be common to (3.16), this is only because of cancellation, as can be seen when directly applying the decomposition of the VWP $_7F_6(1)$ series in (3.14) into balanced $_4F_3(1)$ series:

$$\pi^{3}{}_{4}F_{3}\left[\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2};1\right]$$

$$=2L^{3}{}_{4}F_{3}\left[\frac{1}{4},\frac{1}{4},\frac{1}{4},\frac{1}{4},\frac{1}{4};1\right] -\frac{1}{2}\pi^{2}{}_{4}F_{3}\left[\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2};1\right]$$

$$+\frac{1}{2}\pi^{2}{}_{4}F_{3}\left[\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2};1\right] -\left(\frac{\pi}{L}\right)^{3}{}_{4}F_{3}\left[\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4};1\right].$$
(3.17)

Consideration of which VWP $_7F_6(1)$ series can potentially support the latter $_4F_3(1)$ series in (3.16) in their decomposition into $_4F_3(1)$ series leads to the following two-term relation, which supports our final result on the representation of $\mathscr{W}_4(-1)$ and related series. This provides a remarkable connection between the series involved in (3.5) and (3.6), and a balanced $_5F_4(1)$ series. More expressions involving entries in $\frac{1}{4}$ N and $\frac{1}{8}$ N are provided in the companion paper by McCrorie (2020a).

THEOREM 3.

(a)
$$W\left(1;\frac{1}{2},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4}\right) = \frac{3\pi}{8}W\left(\frac{3}{4};\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{3}{4}\right)$$
 (3.18)

(b)
$$\frac{8\pi^{3}}{L^{4}}{}_{5}F_{4}\left[\frac{\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},1}{\frac{5}{4},\frac{5}{4},\frac{5}{4},\frac{5}{4},\frac{5}{4}};1\right] = \pi^{2}{}_{4}F_{3}\left[\frac{\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}}{\frac{3}{4},\frac{1}{2},\frac{5}{2}};1\right] - \left(2\frac{\pi}{L}\right)^{3}{}_{4}F_{3}\left[\frac{\frac{3}{4},\frac{3}$$

$$\mathcal{W}_{4}(-1) = \frac{8}{\pi^{3}} \int_{0}^{1} K(k)^{2} dk$$
(c)
$$= \frac{\pi}{4} {}_{7}F_{6} \begin{bmatrix} \frac{1}{2}, \frac{5}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, 1 \end{bmatrix}$$

$$= {}_{4}F_{3} \begin{bmatrix} \frac{1}{2}, \frac{1}{2},$$

17

Again, the result is seen fundamentally to be a relation between VWP $_7F_6(1)$ series but the key to (3.19) is that the series $W(1; \frac{1}{2}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4})$ is a $_5F_4(1)$ series in disguise. Closed-form results in the context of $_4F_3(1)$ and $_5F_4(1)$ series are sparse in the literature and, while recent progress has been made by Campbell, D'Aurizio and Sondow (2019), none of the series in Theorems 1 to 3 is currently known in closed form. Nevertheless, we now demonstrate that, at least in principle, the approach followed in this paper *does* potentially offer progress. In the context of constructing the hypergeometry of the parbelos, we show that finding transformed series with entries in $\frac{1}{4}$ N and $\frac{1}{8}$ N provides a route towards finding the closed form of a relevant generalized hypergeometric series. Our approach adds to the four approaches to the characterization problem offered by Campbell, D'Aurizio and Sondow (2020).

4 Hypergeometry of the Parbelos

Sondow (2013) formally introduced the *parbelos* as a parabolic analogue of the arbelos, a classical shape bounded by three pairwise tangent semicircles with collinear diameters. The parbelos corresponding to a given arbelos is obtained by replacing the semicircles of the arbelos with the latus rectum arcs of parabolas opening in the same direction, the foci of which are the centres of the semicircles of the arbelos. Sondow showed that the ratio of the length of the boundary of a parbelos to the length of the corresponding arbelos is always P/π , where P is the *universal parabolic constant*. For any parabola, this constant is the ratio of the arc length of the parabolic segment formed by the latus rectum to the focal parameter. It is given by

$$P = \sqrt{2} + \log\left(1 + \sqrt{2}\right) \approx 2.29558\ 71493\ 92638\ 07403\ 42980\ 49189\ 4903... \tag{4.1}$$

The problem of characterizing the hypergeometry of the parbelos is to find generalized hypergeometric series that sum to P/π . We could say, given the discussion in Section 2.3, that the problem more generally seeks to find allied series up to multiplication by a rational number, or even multiplication by a product or ratio of gamma factors. Campbell, D'Aurizio and Sondow (2020) recently outlined methods in which generalized hypergeometric series relating to the universal parabolic constant could be derived, their first establishing that

$${}_{3}F_{2}\begin{bmatrix}-\frac{1}{2},\frac{1}{4},\frac{3}{4}\\\frac{1}{2},1\end{bmatrix} = \frac{P}{\pi}.$$
(4.2)

Here, we will use the duality concept discussed earlier to derive this series based on a *decomposition* of the basis constant $\log(1 + \sqrt{2})$. The method can also be applied, in principle, to characterize the *equilateral hyperbolic constant*, which is the constant that, for any equilateral (or rectangular) hyperbola, i.e. one whose semi-axes are similar, is the ratio of the area of the latus rectum segment to the square of its semi-axis. It is given by

$$H = \sqrt{2} - \log\left(1 + \sqrt{2}\right) \approx 0.53283\ 99753\ 53552\ 02356\ 90793\ 99229\ 9057...(4.3)$$

We will also show how deriving series with entries in $\frac{1}{8}$ N drives this characterization.

A classical result by Watson (1918) in a mildly reparametrized form states that

$$\frac{\Gamma(a)\Gamma(b)}{\Gamma(a+b)}{}_{3}F_{2}\left[\begin{array}{c}a,b,\frac{a+b-1}{2}\\a+b,\frac{a+b+1}{2}\end{array};1\right] = \frac{1}{2}\left(\psi\left(\frac{a+1}{2}\right) + \psi\left(\frac{b+1}{2}\right) - \psi\left(\frac{a}{2}\right) - \psi\left(\frac{b}{2}\right)\right),\qquad(4.4)$$

where $\psi(x) \coloneqq d\log \Gamma(z)/dz = \Gamma'(x)/\Gamma(x)$ is the digamma function. Kölbig (1996) shows that

$$\psi\left(\frac{p}{q}\right) = -\gamma -\log\left(q\right) + \sum_{j=1}^{q-1} \exp\left(-2\pi i j p/q\right) \log\left(1 - \exp\left(2\pi i j/q\right)\right), \quad (4.5)$$

where $\varepsilon_j = \exp(2\pi i j/q)$ $(j = 0, 1, \ldots, q-1)$ are the q-th roots of unity. This expression motivates taking special values of logarithms of the *cyclotomic polynomials* (polynomials fundamentally factorized using the roots of unity) as basis constants for the class of ${}_{3}F_{2}(1)$ generalized hypergeometric series. See also Ablinger, Blümlein and Schneider (2011, 2013).

McCrorie (2020c) argued that closed-form summation in ${}_{3}F_{2}(1)$ series involving P or H could be naturally seen from the *initial standpoint* of Watson's result, taking $a = b = \frac{1}{4}$ and, at first, using known values of $\{\psi(k/8), 1 \le k < 8, k \in \mathbb{N}\}$, which involve the basis constant $\log(1 + \sqrt{2})$. For example,

$$\psi(1/8) = -\gamma - \frac{\pi}{2} \left(\sqrt{2} + 1\right) - 4\log(2) - \sqrt{2}\log\left(1 + \sqrt{2}\right),\tag{4.6}$$

$$\psi(5/8) = -\gamma + \frac{\pi}{2} \left(\sqrt{2} - 1\right) - 4\log(2) + \sqrt{2}\log\left(1 + \sqrt{2}\right), \tag{4.7}$$

where $\gamma := \lim_{n \to \infty} \left(\sum_{k=1}^{n} \frac{1}{k} - \log(n) \right)$ is the Euler-Mascheroni constant. Application of (4.4) gives upon simplification of the gamma factors the implicit three-term relation

$${}_{3}F_{2}\begin{bmatrix}\frac{1}{4},\frac{1}{4},-\frac{1}{4}\\\frac{1}{2},\frac{3}{4};1\end{bmatrix} = \frac{1}{L}\left(\frac{\pi}{2} + \log\left(1+\sqrt{2}\right)\right).$$
(4.8)

From this base, allied generalized hypergeometric series can be constructed taxonomically using transformations applied to the left-hand-side of (4.8) through various methods. McCrorie (2020c) shows that

$${}_{3}F_{2}\left[\frac{\frac{1}{4},\frac{1}{4},-\frac{1}{4}}{\frac{1}{2},\frac{3}{4}};1\right] = \frac{\sqrt{2}}{L}{}_{3}F_{2}\left[\frac{\frac{1}{4},\frac{1}{2},1}{\frac{3}{4},\frac{5}{4}};1\right] = \frac{1}{L}\left(\frac{\pi}{2} + \frac{\sqrt{2}}{3}{}_{3}F_{2}\left[\frac{\frac{1}{2},1,1}{\frac{5}{4},\frac{7}{4}};1\right]\right)$$

and

$${}_{3}F_{2}\left[\frac{\frac{1}{2},1,1}{\frac{5}{4},\frac{7}{4}};1\right] = \frac{\pi}{2} {}_{3}F_{2}\left[\frac{\frac{1}{4},\frac{1}{2},\frac{3}{4}}{\frac{3}{2},1};1\right].$$

Campbell, D'Aurizio and Sondow (2020) showed that

$${}_{3}F_{2}\begin{bmatrix}\frac{1}{4}, \frac{1}{2}, \frac{3}{4}\\ \frac{3}{2}, 1\end{bmatrix} = \frac{4}{\pi}\ln\left(1 + \sqrt{2}\right),\tag{4.9}$$

and that (4.9) is contiguous to (4.2). The hypergeometry of the parbelos is therefore established from the natural base position given by (4.8).

McCrorie (2020c) motivates the consideration of $\{\psi(k/8), 1 \le k < 8, k \in \mathbb{N}\}$ through a reduction of series with entries in $\frac{1}{4}$ N to series with entries in $\frac{1}{8}$ N, in the spirit of Section 3. He derived the following relationship which he argued should be viewed as fundamental:

$$\sqrt{2}\ln\left(1+\sqrt{2}\right) = {}_{3}F_{2}\left[\frac{\frac{1}{4},\frac{3}{4},1}{\frac{5}{4},\frac{3}{2};1}\right] = \frac{8}{1\times5} {}_{3}F_{2}\left[\frac{\frac{1}{8},\frac{5}{8},1}{\frac{9}{8},\frac{13}{8};1}\right] - \frac{8}{3\times7} {}_{3}F_{2}\left[\frac{\frac{3}{8},\frac{7}{8},1}{\frac{11}{8},\frac{15}{8};1}\right].$$
(4.10)

The two ${}_{3}F_{2}(1)$ series on the right-hand-side are free of Pochhammer symbols, viz.

$$\frac{1}{5}{}_{3}F_{2}\left[\frac{\frac{1}{8},\frac{5}{8},1}{\frac{9}{8},\frac{13}{8}};1\right] = \sum_{k=0}^{\infty} \frac{1}{(8k+1)(8k+5)},$$
(4.11)

and

$$\frac{1}{21}{}_{3}F_{2}\left[\frac{\frac{3}{8},\frac{7}{8},1}{\frac{11}{8},\frac{15}{8}};1\right] = \sum_{k=0}^{\infty} \frac{1}{(8k+3)(8k+7)}.$$
(4.12)

With entries in $\frac{1}{8}$ N, the series are therefore of a form that is summable, up to a constant that is the product of a rational number and ϖ_8 or its reciprocal, *in terms of* the digamma functions { $\psi(k/8)$, $1 \le k < 8$, $k \in \mathbb{N}$ } (see Al-Saqabi, Kalla and Srivastava, 1991). The method supports an approach to series summation whose evaluation involves the logarithm of $1 + \sqrt{2}$, and extrapolates via (4.4) and (4.5) to a wider class of basis constants based on the cyclotomic polynomials. Ablinger, Blümlein and Schneider (2011, 2013) set this idea in the context of sums relevant to Mathematical Physics. As noted, the issue of finding (or defining) basis constants for the series in Section 3 is more challenging because the underlying series are of $_4F_3(1)$ type.

5 An Analogous Problem in Econometrics

We now establish a connection between a classical, still-open problem in Econometrics and VWP $_7F_6(1)$ series. Consider the first-order autoregressive, AR(1), process⁸

$$x_t = \rho x_{t-1} + \varepsilon_t (t = 1, ..., T)$$
(5.1)

where $\varepsilon_t \sim NID(0, 1)$, $x_0 = 0$ and $|\rho| = 1$. Although (5.1) is highly idealized from today's standpoint, it is canonical as a building block for more useful and realistic econometric time series models in a sense explicitly outlined by Abadir (1992). The ordinary least squares (OLS) estimator of ρ based on a sample of size T coincides with the Gaussian maximum likelihood estimator and is given by

$$\widehat{\rho}_T = \sum_{t=1}^T x_t x_{t-1} / \sum_{t=1}^T x_{t-1}^2.$$
(5.2)

 $^{^{8}}$ The material in this section is based on the framework adopted by McCrorie (2020b).

In contrast to the uniform random walk problem, where in the multidimensional case each step corresponds to a random vector distributed on the unit sphere, here it is the model, parametrized by ρ , which nests a random walk (along with other types of behaviour). Our objective is to estimate this parameter on the basis of data and, in this context, it is the *estimator*, $\hat{\rho}_T$, that connects to the hypersphere through its being a ratio of quadratic forms.⁹ Clearly, the properties of the estimator $\hat{\rho}_T$ vary across different regions in the parameter space. The characterization of, for example, the bias of the estimator,

$$b_T(\rho) \coloneqq E\left(\widehat{\rho}_T\right) - \rho, \qquad (5.3)$$

has been a classical problem in Econometrics, see e.g. Hurwicz (1950), White (1961) and Shenton and Johnson (1965). As $T \rightarrow \infty$,

If
$$|\rho| < 1$$
, $Tb_T(\rho)$ converges in probability to -2ρ ;
If $\rho = \pm 1$, $Tb_T(\rho)$ converges to $\pm m \rho$, where *m* is a negative constant;
If $|\rho| > 1$, $T^{-1/2} |\rho|^T b_T(\rho)$ converges to $-2^{-1/2} \pi^{1/2} \rho^{-1} (\rho^2 - 1)^{3/2}$.
(5.4)

See also Le Breton and Pham (1989) and Abadir (1993a). The problem of characterizing the constant m appearing in (5.4), which is *negative* in the positive parameter case (and positive in the negative case), remains open. An equivalent way of expressing the problem is that m is the mean of the asymptotic distribution of the test statistic under the unit root null hypothesis $\rho = 1$. This distribution, called the Dickey-Fuller distribution after Dickey and Fuller (1979, 1981), has become pervasive in the area of econometric time series analysis,¹⁰ a discipline whose underlying (economic) variables seemingly exhibit trending behaviour. It is surprising that, forty years on, little is known about its mean when viewed as a mathematical constant. Renewed interest in the constant has followed from work on predictive regression by Phillips (2012, 2015). The model is typified by the specification

$$y_t = \beta x_{t-1} + u_{0t}, \tag{5.5}$$

$$x_t = \rho x_{t-1} + u_{xt}, (5.6)$$

⁹ See Hillier (2001), Forchini (2002) and references therein.

¹⁰ See also Rao (1978), Phillips (1987) and Abadir (1993b). Tanaka (2017) provides a textbook treatment.

where the objective is to predict the scalar time series y_t given past information embodied in a set of regressors x_{t-1} under an assumption such as (u_t, \mathcal{F}_t) is a martingale difference sequence with

$$E(u_{t}u_{t}'|\mathcal{F}_{t-1}) = \begin{bmatrix} \sigma_{00} & \sigma_{0x} \\ \sigma_{x0} & \Sigma_{xx} \end{bmatrix},$$

where $\mathcal{F}_{t-1} = \sigma(u_t, u_{t-1}, \ldots)$ is the natural filtration associated with innovations $u_t = (u_{0t}, u_{xt})'$. Applying OLS to (5.5) under the assumption that x_t is scalar, and setting $u_{0,xt} = u_{0t} - \sigma_{0x} \Sigma_{xx}^{-1} u_{xt}$, the estimation error decomposes as

$$\widehat{\beta} - \beta = \frac{\sum_{t=1}^{T} x_{t-1} u_{0.xt}}{\sum_{t=1}^{T} x_{t-1}^2} + \sigma_{0x} \Sigma_{xx}^{-1} \Big(\widehat{\rho} - \rho \Big),$$
(5.8)

where $\widehat{\rho} = \left(\sum_{t=1}^{T} x_{t-1}^2\right)^{-1} \sum_{t=1}^{T} x_{t-1} x_t$. Taking expectations,

$$E\left(\widehat{\beta}-\beta\right) = \sigma_{0x} \Sigma_{xx}^{-1} E\left(\widehat{\rho}-\rho\right) = \sigma_{0x} \Sigma_{xx}^{-1} B_T(\rho), \qquad (5.9) \quad (5.7)$$

where the autoregressive bias function

$$B_T(\rho) \coloneqq E\left(\widehat{\rho} - \rho\right) \tag{5.10}$$

depends only on ρ and T. Phillips (2012) provides an exact formula for $B_T(\rho)$ under Gaussianity and the following complete set of large-T asymptotic expansions

$$B_{T}(\rho) = \begin{cases} \frac{-2\rho/T + O(T^{-2})}{m/T + O(T^{-2})} & |\rho| < 1\\ m/T + O(T^{-2}) & \rho = 1\\ -g(c)/T + O(T^{-2}) & \rho = 1 + c/T,\\ O(1/|\rho|^{T}) & |\rho| > 1 \end{cases}$$
(5.11)

where g(c) is a continuous function of a constant c given explicitly by Phillips (2012) and $\lim_{c \to 0} g(c) = -m$.

McCrorie (2020b) classified a set of generalized hypergeometric series that are allied to the constant m, defining the canonical member of the set to be

$$M \coloneqq 1 - m = {}_{3}F_{2} \left[\frac{\frac{1}{2}}{\frac{3}{2}}, \frac{1}{5}, \frac{1}{4}; 1 \right] \cong 2.78143\ 01712\ 77837\ 42085\ 88648\ 61173\ 56... \tag{5.12}$$

Treated as basis constants, M and another constant constructed in parallel¹¹ were seen along with π and L to be involved in an expression for a critical value

¹¹ This constant relates to M in precisely the way $\Gamma(\frac{1}{4})$ relates to $\Gamma(\frac{3}{4})$, and the series $W(\frac{1}{4}; \frac{1}{8}, \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{5}{8})$ and $W(\frac{3}{4}; \frac{3}{8}, \frac{1}{2}, \frac{1}{2}, \frac{3}{4}, \frac{7}{8})$ in Theorem 1 relate to each other.

of the *L*-series of an elliptic curve of conductor 32, one of whose isogenies is the "congruent number elliptic curve" (Tian, 2015) and another which has the modular form parametrization $f_{2, 32}(\tau) = \eta(4\tau)^2\eta(8\tau)^2$. They are LMFDB elliptic curves 32.a3 and 32.a4 respectively.¹² See Rodriguez Villegas (1999), Zudilin (2013) and Ito (2018) for a discussion of this *L*-series evaluation problem from the point of view of Analytic Number Theory.

We now show that, like the moments in the four-step uniform random walk problem, the constant M is representable in terms of VWP $_7F_6(1)$ series. This is facilitated by the following lemma which involves a transformation to a $_3F_2(1)$ series that is allied to the series in (5.12).

LEMMA 4.

$${}_{3}F_{2}\left[\frac{\frac{3}{4},1,1}{\frac{3}{2},\frac{7}{4}};1\right] = 3\left(M - \frac{\pi}{4}L\right).$$
(5.13)

THEOREM 5.

$$M - \frac{\pi}{4}L = \frac{1}{3}W\left(\frac{3}{2}; \frac{3}{4}, \frac{3}{4}, 1, 1, \frac{5}{4}\right)$$
(5.14)

$$= \frac{1}{2} \left(\frac{\pi}{L}\right)^2 W\left(\frac{1}{2}; \frac{1}{4}, \frac{1}{4}, \frac{1}{4}, \frac{1}{2}, \frac{1}{2}\right)$$
(5.15)

$$= \frac{3}{16} \frac{\pi^2}{L} W\left(\frac{3}{4}; \frac{1}{4}, \frac{1}{4}, \frac{1}{2}, \frac{3}{4}, \frac{3}{4}\right)$$
(5.16)

$$=\frac{2}{3}W\left(1;\frac{1}{4},\frac{1}{2},\frac{3}{4},\frac{3}{4},1\right)$$
(5.17)

 $^{{}^{12}}f(\tau) \coloneqq f_{2,32} \text{ is an example of a type of multiplicative } \eta \text{-} \text{product discussed by Voskresenskaya} (2012) \text{ such that } f^2(\tau/2) \text{ is also a multiplicative } \eta \text{-} \text{product. Here, } f^2(\tau/2) = \eta(2\tau)^4 \eta(4\tau)^4 = f_{4,8}.$

$$=\frac{5}{9}W\left(\frac{5}{4};\frac{1}{2},\frac{1}{2},1,1,1\right)$$
(5.18)

$$=\frac{5L}{24}W\left(\frac{5}{4};\frac{1}{2},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{5}{4}\right)$$
(5.19)

The expressions can be read in two ways. On one hand, M and so m are expressed in terms of a number of VWP $_7F_6(1)$ series. On the other, if Mis taken to be a basis constant, the series themselves, which are not symbolically evaluated in term of known constants by software such as *Mathematica 11.2*, can be expressed in closed form within the Q-linear span of $\{\pi, L, M\}$. McCrorie (2020b) shows that each $_7F_6(1)$ series can be split-up into balanced $_4F_3(1)$ series with entries in $\frac{1}{4}$ N and $\frac{1}{8}$ N, and closed form expressions for such series involving a wider class of constants including M can be obtained. These evaluations complement those that were obtained for allied $_3F_2(1)$ series.

6 Conclusion

This paper has provided new results and perspective on a mathematical constant that appears in the representation of the odd moments in even dimensions of Pearson's four-step uniform random walk problem. The symmetries in (3.11), (3.12), (3.15), (3.16) and (3.20) formally reveal the sense in which the constant defined by the series (3.2) has a counterpart in the generalized hypergeometric series in (3.5). As described above, both constants are manifest in a variety of fields. While the question of whether either constant can be reduced to a Q-linear span of a more fundamental set of constants (known or to be defined) remains open, relating both constants to VWP $_7F_6(1)$ series with entries in $\frac{1}{4}$ N and $\frac{1}{8}$ N allowed a number of series of lower order to be related to them. While the results are of interest in their own right, they also open up the possibility of establishing the constants in terms of more elemental closed forms following the approach taken by McCrorie (2020c) to the universal parabolic constant summarized in Section 4.

The paper also motivates questions for possible future work. By providing a formal sense in which the two constants in Section 3 were counterparts of each

25

other, the paper goes farther than Rogers, Wan and Zucker (2016) in describing how the critical values of the *L*-series of two related, even-weight cusp forms are intertwined. It would be interesting to investigate whether the type of the symmetries exhibited here apply more widely in the interplay between modular forms and generalized hypergeometric series and, if so, whether they can be built up in a similar way. Recent results by Straub and Zudilin (2020) and Brunault and Zudilin (2020) also suggest a connection with Mahler measure.

Given for any weight 4 modular form f, the functional equation relates L(f, s) and L(f, 4-s), the value at s = 4 may have special importance. Papanikolas, Rogers and Samart (2014) and Wan and Zucker (2016) have derived hypergeometric series expressions for the *L*-series of $f_{4, 8}$ evaluated at 4, one being

$$L(f_{4,8},4) = \frac{\pi^4}{192} \left\{ {}_5F_4 \left[\begin{array}{c} \frac{1}{2}, \frac{1}{2},$$

where $\zeta(s) \coloneqq \sum_{n=1}^{\infty} \frac{1}{n^s} (\operatorname{Re}(s) > 1)$ is the Riemann zeta function. It would be interesting to explore whether (6.1) encodes any information beyond that

encoded in the critical values discussed here. A generalized hypergeometric series representation for $L(f_{4, 16}, 4)$ could also be derived and its relation to (6.1) explored.

Another direction for future work might relate the results in the paper more strongly to the property that the densities of the short-walk problems are recursively related. Indeed, following Hughes (1995), if $\phi_N(x) = p_N(x)/2\pi x$, then for integers $N \ge 2$,

$$\phi_N(x) = \frac{1}{2\pi} \int_0^{2\pi} \phi_{N-1} \left(\sqrt{x^2 - 2x \cos \alpha + 1} \right) d \alpha.$$
 (6.2)

While this aspect has already been exploited, the underlying complexity of the short-walk problem for small N has meant that tractable expressions have still been difficult to obtain. On the other hand, some sharp results have been obtained on certain special values of the density functions. For example, Borwein, Straub, Wan and Zudilin (2012) and Joyce (2017) show using different methods that

$$p_4(1) = p'_5(0) = \frac{1}{8\pi^4\sqrt{5}} \Gamma\left(\frac{1}{15}\right) \Gamma\left(\frac{2}{15}\right) \Gamma\left(\frac{4}{15}\right) \Gamma\left(\frac{8}{15}\right) = \frac{3\sqrt{5}}{\pi^3} \frac{(\sqrt{5}-1)}{2} K_{15}^2, \quad (6.3)$$

where K_{15} is the complete elliptic integral evaluated at the 15-th singular value. The five-step random walk problem centres around a weight 3 level 15 cusp form

$$f_{3,15}(\tau) = [\eta(\tau)\eta(15\tau)]^3 + [\eta(3\tau)\eta(5\tau)]^3.$$
(6.4)

See especially Bloch, Kerr and Vanhove (2015), Samart (2016) and Zhou (2019a). This context offers another instance of the observation of Rogers, Wan and Zucker (2015) of an *L*-series of an odd-weighted modular form having critical values that are products of gamma functions. Because the functional equation here relates $L(f_{3,15}, s)$ and $L(f_{3,15}, 3-s)$, all the critical values relate to each other in a way that avoids intertwinement with another modular form as was exemplified here by (3.5) and (3.6). Specifically,

$$\frac{\pi^2}{15}L(f_{3,15},1) = \frac{1}{2}\sqrt{\frac{\pi^2}{15}}L(f_{3,15},2) = \frac{\pi^4}{30}\int_0^\infty J_0(t)^4 dt = \int_0^\infty tI_0(t)K_0(t)^4 dt = \frac{1}{240\sqrt{5}}\Gamma\left(\frac{1}{15}\right)\Gamma\left(\frac{2}{15}\right)\Gamma\left(\frac{4}{15}\right)\Gamma\left(\frac{8}{15}\right).$$
(6.5)

The ratio of the above value to π^2 is the so-called *Bologna constant* introduced by Laporta (2008) *op. cit.* and independently by Broadhurst (2007) in work that was developed in the jointly authored paper by Bailey, Borwein, Broadhurst and Glasser (2008). Characterizing the commonalities and differences in the types of expressions for moments that relate to even versus odd weight modular forms remains an important open problem.

Last, but by no means least, this paper has provided a connection between moment properties of the distribution of a short-walk problem and the Dickey-Fuller distribution that is now pervasive in Econometrics. The distribution theory underlying statistics like (5.2) that are ratios of quadratic forms is still incompletely characterized in the asymptotic case, and is less resolved in the finite sample case (see e.g. Hillier, 2001; Forchini, 2002; and references therein). Very few analytical expressions are available for the relevant densities and distributions beyond what K.M. Abadir derived in a number of papers in the 1990s (see, e.g. Abadir, 1993b; Abadir and Lucas, 2004). In spite of work by Van Garderen (1999, 2000), the underlying geometry of non-stationary autoregressive models is still not fully characterized. The shared characteristics between the four-step random walk problem and least-squares estimation in autoregressive models could motivate new approaches towards the latter that embrace machinery currently commonplace in Analytic Number Theory and Mathematical Physics, and more systematically encompass numerical and experimental methods. The fundamental problem in autoregressive models, in the absence of a well-developed finite-sample theory, is the polarization in their properties that manifests itself in large samples. Some recent methods seek to confront this problem directly. The IVX endogenous instrumentation procedure introduced by Phillips and Magdalinos (2009) and developed by Phillips and Lee (2013, 2016), Kostakis, Magdalinos and Stamatogiannis (2015) and Phillips (2015) is central to this effort.

The intention of writing the current paper was to use the setting of the fourstep random walk problem to offer a confluence of ideas around the constant represented by the VWP $_7F_6(1)$ series in (3.2) involved in the representation of the odd moments in even dimensions. The approach underpinned a multidisciplinary perspective of a type encouraged by the journal's founding editor, P.C. Mahalanobis. The *MacTutor History of Mathematics Archive* – a resource that provides, *inter alia*, summaries of the historical contribution of important mathematicians in the last few centuries – is held in the School of Mathematical Sciences in the author's current institution, the University of St Andrews. While not a substitute for Rao's (1973) important biographical article or the biography by Rudra (1996), its summary entry for Prasanta Chandra Mahalanobis can be accessed here: http://www-groups.dcs.st-and.ac. uk/history/Biographies/Mahalanobis.html.

Acknowledgement I am grateful for the comments of two anonymous referees and participants at the *EcoSta 2019* conference held at the National Chung Hsing University, Taiwan, 25-27 June 2019.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

ABADIR, K.M. (1992) A DISTRIBUTION GENERATING EQUATION FOR UNIT ROOT STATISTICS. OXFORD BULLETIN OF ECONOMIC STATISTICS 54, 305-323.

ABADIR, K.M. (1993A) OLS BIAS IN A NONSTATIONARY AUTOREGRESSION. ECONOMETRIC THEORY 9, 81-93.

- ABADIR, K.M. (1993b) THE LIMITING DISTRIBUTION OF THE AUTOCORRELATION COEFFICIENT UNDER A UNIT ROOT. ANNALS OF STATISTICS 21(2), 1058-1070.
- ABADIR, K.M. AND LUCAS, A. (2004) A COMPARISON OF MINIMUM MSE AND MAXIMUM POWER FOR THE NEARLY INTEGRATED NON-GAUSSIAN MODEL. JOURNAL OF ECONOMETRICS 119, 45-71.
- ABLINGER, J. AND BLÜMLEIN, J. (2013) HARMONIC SUMS, POLYLOGARITHMS, SPECIAL NUMBERS, AND THEIR GENERALIZATIONS. IN C. SCHNEIDER AND J. BLÜMLEIN (EDS), COMPUTER ALGEBRA IN QUANTUM FIELD THEORY, TEXTS & MONOGRAPHS IN SYMBOLIC COMPUTATION, PP. 1-32. SPRINGER.
- ABLINGER, J., BLÜMLEIN, J. AND SCHNEIDER, C. (2011) HARMONIC SUMS AND POLYLOGARITHMS GENERATED BY CYCLOTOMIC POLYNOMIALS. JOURNAL OF MATHEMATICAL PHYSICS 52, 102301 (52PP.)
- ABLINGER, J., BLÜMLEIN, J. AND SCHNEIDER, C. (2013) ANALYTIC AND ALGORITHMIC ASPECTS OF GENERALIZED HARMONIC SUMS AND POLYLOGARITHMS. JOURNAL OF MATHEMATICAL PHYSICS 54, 082031 (74pp.)
- AL-SAQABI, B.N., KALLA, S.L. AND SRIVASTAVA, H.M. (1991) A CERTAIN FAMILY OF INFINITE SERIES ASSOCIATED WITH DIGAMMA FUNCTIONS. JOURNAL OF MATHEMATICAL ANALYSIS AND APPLICATIONS 159, 361-372.
- BAILEY, D.H., BORWEIN, J.M., BROADHURST, D. AND GLASSER, M.L. (2008) ELLIPTIC INTEGRAL EVALUATIONS OF BESSEL MOMENTS AND APPLICATIONS. JOURNAL OF PHYSICS A: MATHEMATICAL AND THEORETICAL 41(20), 5203-5231.

BAILEY, D.H. AND BROADHURST, D.J. (2000) PARALLEL INTEGER RELATION DETECTION: TECHNIQUES AND APPLICATIONS. *MATHEMATICS OF COMPUTATION* 70, 1719-1736.

- BAILEY, W.N. (1935) *GENERALIZED HYPERGEOMETRIC SERIES*. CAMBRIDGE TRACTS IN MATHEMATICS AND MATHEMATICAL PHYSICS, NO. 32. CAMBRIDGE UNIVERSITY PRESS.
- BLOCH, S., KERR, M. AND VANHOVE, P. (2015) A FEYNMAN INTEGRAL VIA HIGHER NORMAL FUNCTIONS. COMPOSITIO MATHEMATICA 151(12), 2329-2375.
- BORWEIN, J.M. (2016) A SHORT WALK CAN BE BEAUTIFUL. JOURNAL OF HUMANISTIC MATHEMATICS 6(1), 86-109.
- BORWEIN, J.M. AND CRANDALL, R.E. (2013) CLOSED FORMS: WHAT THEY ARE AND WHY WE CARE? NOTICES OF THE AMERICAN MATHEMATICAL SOCIETY **60**, 50-65.
- BORWEIN, J.M., NUYENS, D., STRAUB A. AND WAN, J. (2011) SOME ARITHMETIC PROPERTIES OF RANDOM WALK INTEGRALS. THE RAMANUJAN JOURNAL **26(1)**, 109-132.
- BORWEIN, J.M. AND SINNAMON, C. W. (2016) A CLOSED FORM FOR THE DENSITY FUNCTIONS OF RANDOM WALKS IN ODD DIMENSIONS. BULLETIN OF THE AUSTRALIAN MATHEMATICAL SOCIETY 93(2), 330-339.
- BORWEIN, J.M. AND STRAUB, A. (2013) MAHLER MEASURES, SHORT WALKS AND LOG-SINE INTEGRALS. THEORETICAL COMPUTER SCIENCE 479, 4-21.
- BORWEIN, J.M., STRAUB, A. AND VIGNAT, C. (2016) DENSITIES OF SHORT UNIFORM RANDOM WALKS IN HIGHER DIMENSIONS. JOURNAL OF MATHEMATICAL ANALYSIS AND ITS APPLICATIONS 437(1), 668-707.
- BORWEIN, J.M., STRAUB, A. AND WAN, J. (2013) THREE-STEP AND FOUR-STEP RANDOM WALK INTEGRALS. EXPERIMENTAL MATHEMATICS 22(1), 1-14.
- BORWEIN, J.M., STRAUB, A., WAN, J. AND ZUDILIN, W. (2012) DENSITIES OF SHORT UNIFORM RANDOM WALKS. CANADIAN JOURNAL OF MATHEMATICS 64(5), 961-990. (WITH AN APPENDIX BY D. ZAGIER).

- BROADHURST, D. (2007) RECIPROCAL PSLQ AND THE TINY NOME OF BOLOGNA. INTERNATIONAL WORKSHOP IN FRONTIERS IN PERTURBATIVE QUANTUM FIELD THEORY, ZENTRUM FÜR INTERDISZIPLINÄRE FORSCHUNG IN BIELEFELD, JUNE 14 2007. AVAILABLE AT: http://www.physik.uni-bielefeld. DE/IGS/SCHOOLS/ZIF2007/BROADHURST.PDF.
- BROADHURST, D. (2013) MULTIPLE ZETA VALUES AND MODULAR FORMS IN QUANTUM FIELD THEORY. IN C. SCHNEIDER AND J. BLÜMLEIN (EDS), COMPUTER ALGEBRA IN QUANTUM FIELD THEORY, TEXTS & MONOGRAPHS IN SYMBOLIC COMPUTATION, PP. 33–73. SPRINGER.
- BROADHURST, D. (2016) FEYNMAN INTEGRALS, *L*-SERIES AND KLOOSTERMAN MOMENTS. COMMUNICATIONS IN NUMBER THEORY AND PHYSICS **10(3)**, 527-569.
- BROADHURST, D. AND A. MELLIT (2016) PERTURBATIVE QUANTUM FIELD THEORY INFORMS ALGEBRAIC GEOMETRY. LOOPS AND LEGS IN QUANTUM FIELD THEORY 2016. PROCEEDINGS OF SCIENCE 260, 079 (8PP.). https://doi.org/10.22323/1.260.0079.
- BROADHURST, D. AND D.P. ROBERTS (2018) QUADRATIC RELATIONS BETWEEN FEYNMAN INTEGRALS. LOOPS AND LEGS IN QUANTUM FIELD THEORY 2018, *PROCEEDINGS OF SCIENCE* **303**, 053 (9PP). HTTPS://DOI.ORG/10.22323/1.303.0053.
- BRUNAULT, F. AND W. ZUDILIN (2020) MANY VARIATIONS OF MAHLER MEASURES: A LASTING SYMPHONY AUSTRALIAN MATHEMATICAL SOCIETY LECTURE SERIES 28, CAMBRIDGE UNIVERSITY PRESS.
- CAMPBELL, J.M., D AURIZIO, J. AND SONDOW, J. (2019) ON THE INTERPLAY AMONG HYPERGEOMETRIC FUNCTIONS, COMPLETE ELLIPTIC INTEGRALS AND FOURIER-LEGENDRE EXPANSIONS. *JOURNAL OF MATHEMATICAL ANALYSIS AND APPLICATIONS* **479(1)**, 90-121.
- CAMPBELL, J.M., D AURIZIO, J. AND SONDOW, J. (2020) HYPERGEOMETRY OF THE PARBELOS. THE AMERICAN MATHEMATICAL MONTHLY 127(1), 23-32.
- DICKEY, D.A. AND FULLER, W.A. (1979) DISTRIBUTION OF THE ESTIMATORS FOR AUTOREGRESSIVE TIME SERIES WITH A UNIT ROOT. JOURNAL OF THE AMERICAN STATISTICAL ASSOCIATION 74, 427-431.
- DICKEY, D.A. AND FULLER, W.A. (1981) LIKELIHOOD RATIO STATISTICS FOR AUTOREGRESSIVE TIME SERIES WITH A UNIT ROOT. ECONOMETRICA 49, 1057-1072.
- FORCHINI, G. (2002) THE EXACT CUMULATIVE DISTRIBUTION FUNCTION OF A RATIO OF QUADRATIC FORMS IN NORMAL VARIABLES, WITH APPLICATION TO THE AR(1) MODEL. *ECONOMETRIC THEORY* 18, 823-852.
- GHOSH, J.K. (1994) MAHALANOBIS AND THE ART AND SCIENCE OF STATISTICS: THE EARLY DAYS. INDIAN JOURNAL OF HISTORY OF SCIENCE 29(1), 89-98.
- GUTTMANN, A.J. (2010) LATTICE GREEN'S FUNCTIONS IN ALL DIMENSIONS. JOURNAL OF PHYSICS A: MATHEMATICAL AND THEORETICAL 43, 305205 (26PP).
- HILLIER, G. (2001) THE DENSITY OF A QUADRATIC FORM IN A VECTOR UNIFORMLY DISTRIBUTED ON THE N-SPHERE. ECONOMETRIC THEORY 17, 1-28.
- HUGHES, L. (1995) RANDOM WALKS AND RANDOM ENVIRONMENTS: VOL. 1. RANDOM WALKS. OXFORD UNIVERSITY PRESS.
- HURWICZ, L. (1950) LEAST SQUARES ESTIMATION IN TIME SERIES MODELS. IN T. KOOPMANS (ED.), STATISTICAL INFERENCE IN DYNAMIC ECONOMIC MODELS, PP. 365-383. WILEY.
- ITO, R. (2018) THE BEILINSON CONJECTURES FOR CM ELLIPTIC CURVES VIA HYPERGEOMETRIC FUNCTIONS. THE RAMANUJAN JOURNAL 45(2), 433-449.
- JOYCE, G.S. (2017) EXACT RESULTS FOR THE DIAMOND LATTICE GREEN FUNCTION WITH APPLICATIONS TO UNIFORM RANDOM WALKS IN THE PLANE. JOURNAL OF PHYSICS A: MATHEMATICAL AND THEORETICAL 50, 423001 (29PP).
- KLUYVER, J.C. (1905) A LOCAL PROBABILITY PROBLEM. NEDERLANDSCHE AKADEMIE VAN WETENSCHAPPEN PROCEEDINGS 8(1), 341-350.

- KOBLITZ, N. (1993) INTRODUCTION TO ELLIPTIC CURVES AND MODULAR FORMS. SECOND EDITION. SPRINGER.
- KÖLBIG, K.S. (1996) THE POLYGAMMA FUNCTION AND THE DERIVATIVES OF THE COTANGENT FUNCTION FOR RATIONAL ARGUMENTS. CERN-CN-96-005, CERN COMPUTING AND NETWORKS DIVISION. AVAILABLE AT: HTTP://CDS.CERN.CH/RECORD/298844.
- Kontsevich, M. and Zagier, D. (2001) periods. In B. Engquist and W. Schmid (Eds) *mathematics unlimited* – 2001 and beyond, pp. 771-808. springer.
- KOSTAKIS, A., MAGDALINOS, T. AND STAMATOGIANNIS, M.P. (2015) ROBUST ECONOMETRIC INFERENCE FOR STOCK RETURN PREDICTABILITY. THE REVIEW OF FINANCIAL STUDIES 28(5), 1506-1553.
- LAPORTA, S. (2008) ANALYTICAL EXPRESSIONS OF 3- AND 4-LOOP SUNRISE FEYNMAN INTEGRALS AND 4-DIMENSIONAL LATTICE INTEGRALS. *INTERNATIONAL JOURNAL OF MODERN PHYSICS A* 23(31), 5007-502.
- LAPORTA, S. (2018) BUILDING BASES FOR ANALYTICAL FITS OF FOUR-LOOP QED MASTER INTEGRALS. LOOPS AND LEGS IN QUANTUM FIELD THEORY 2018. *PROCEEDINGS OF SCIENCE* **303**, 073 (8PP.) HTTPS://DOI.ORG/10.22323/1.303.0073.
- LE BRETON, A. AND PHAM, D.T. (1989) ON THE BIAS OF THE LEAST SQUARES ESTIMATOR FOR THE FIRST ORDER AUTOREGRESSIVE PROCESS. ANNALS OF THE INSTITUTE OF STATISTICAL MATHEMATICS 41(2), 555-563.
- MARTIN, Y. (1996) MULTIPLICATIVE η-PRODUCTS. TRANSACTIONS OF THE AMERICAN MATHEMATICAL SOCIETY 348, 4825-4856.
- MCCRORIE, J.R. (2020A) GENERALIZED HYPERGEOMETRIC SERIES ALLIED TO MOMENTS IN PEARSON'S FOUR-STEP UNIFORM RANDOM WALK PROBLEM. IN PREPARATION.
- MCCRORIE, J.R. (2020b) THE MEAN OF THE DICKEY-FULLER DISTRIBUTION VIEWED AS A MATHEMATICAL CONSTANT. IN PREPARATION.
- MCCRORIE, J.R. (2020C) DUALITY IN THE HYPERGEOMETRY OF THE PARBELOS. IN PREPARATION.
- NAYAK, T.K. (2009) IMPACT OF KARL PEARSON'S WORK IN THE DEVELOPMENT OF STATISTICS IN INDIA. INTERNATIONAL STATISTICAL REVIEW 77, 72-80.
- OSBURN, R. AND STRAUB, A. (2019) INTERPOLATED SEQUENCES AND CRITICAL *L*-VALUES OF MODULAR FORMS. IN BLÜMLEIN, J., SCHNEIDER, C. AND PAULE, P. (EDS) *ELLIPTIC INTEGRALS, ELLIPTIC FUNCTIONS AND MODULAR FORMS IN QUANTUM FIELD THEORY*, PP. 327-349. SPRINGER.
- PAPANIKOLAS, M.A., ROGERS, M.D. AND SAMART, D. (2014) THE MAHLER MEASURE OF A CALABI-YAU THREEFOLD AND SPECIAL *L*-VALUES. *MATHEMATISCHE ZEITSCHRIFT* **276**, 1151-1163.
- PEARSON, K. (1905) THE PROBLEM OF THE RANDOM WALK. NATURE 72, 294, 342.
- PHILLIPS, P.C.B. (1987) TIME SERIES REGRESSION WITH A UNIT ROOT. ECONOMETRICA 55, 277-301.
- PHILLIPS, P.C.B. (2012) FOLKLORE THEOREMS, IMPLICIT MAPS AND INDIRECT INFERENCE. ECONOMETRICA 80(1), 425-454.
- PHILLIPS, P.C.B. (2015) PITFALLS AND POSSIBILITIES IN PREDICTIVE REGRESSION. HALBERT WHITE MEMORIAL LECTURE. JOURNAL OF FINANCIAL ECONOMETRICS 13(3), 521-555.
- PHILLIPS, P.C.B. AND LEE, J. H. (2013) PREDICTIVE REGRESSION UNDER VARYING DEGREES OF PERSISTENCE AND ROBUST LONG-HORIZON REGRESSION. JOURNAL OF ECONOMETRICS 177(2), 250-24.
- PHILLIPS, P.C.B. AND LEE, J. H. (2016) ROBUST ECONOMETRIC INFERENCE WITH MIXED INTEGRATED AND MILDLY EXPLOSIVE REGRESSORS. JOURNAL OF ECONOMETRICS 192(2), 433-450.
- PHILLIPS, P.C.B. AND T. MAGDALINOS (2009) ECONOMETRIC INFERENCE IN THE VICINITY OF UNITY. COFIE WORKING PAPER NO. 6, SINGAPORE MANAGEMENT UNIVERSITY.
- PRUDNIKOV, A.P., BRYCHKOV, Y.A. AND MARICHEV, O.I. (1986) INTEGRALS AND SERIES: MORE SPECIAL FUNCTIONS (VOLUME 3). GORDON AND BEACH.
- RAO, C.R. (1973) PRASANTA CHANDRA MAHALANOBIS. 1893-1972. *BIOGRAPHICAL MEMOIRS OF FELLOWS OF THE ROYAL SOCIETY* **19**, 454-492.

- RAO, M.M. (1978) ASYMPTOTIC DISTRIBUTION OF AN ESTIMATOR OF THE BOUNDARY PARAMETER OF AN UNSTABLE PROCESS. ANNALS OF STATISTICS 6(1), 185-190. (CORRECTION: ANNALS OF STATISTICS 8(6), 1403.)
- RICHMOND, L.B. & J. SHALLIT (2009) COUNTING ABELIAN SQUARES. THE ELECTRONIC JOURNAL OF COMBINATORICS 16(1), #R72.
- RAYLEIGH L. (1905) THE PROBLEM OF THE RANDOM WALK. NATURE 72, 318.
- RODRIGUES VILLEGAS, F. (1999) MODULAR MAHLER MEASURES I. IN S.D. AHLGREN, G.E. ANDREWS & K. ONO (EDS) TOPICS IN NUMBER THEORY, PP. 17-48. MATHEMATICS AND ITS APPLICATIONS 467, SPRINGER.
- ROGERS, M., WAN, J.G. AND ZUCKER, I.J. (2015) MOMENTS OF ELLIPTIC INTEGRALS AND CRITICAL *L*-VALUES. *THE RAMANUJAN JOURNAL* **37**, 113-130.
- RUDRA, A. (1996) PRASANTA CHANDRA MAHALANOBIS: A BIOGRAPHY. OXFORD UNIVERSITY PRESS.
- SAMART, D. (2016) FEYNMAN INTEGRALS AND CRITICAL MODULAR *L*-FUNCTIONS. COMMUNICATIONS IN NUMBER THEORY AND PHYSICS 10(1), 133-156.
- SHENTON, L.R. AND N.L. JOHNSON (1965) MOMENTS OF A SERIAL CORRELATION COEFFICIENT. JOURNAL OF THE ROYAL STATISTICAL SOCIETY SERIES B: STATISTICAL AND METHODOLOGICAL 27, 308-320.
- SONDOW, J. (2013) THE PARBELOS, A PARABOLIC ANALOG OF THE ARBELOS. AMERICAN MATHEMATICAL MONTHLY 120, 929-935.
- STRAUB, A. AND W. ZUDILIN (2020) SHORT WALK ADVENTURES. IN D. H. BAILEY, N.S. BORWEIN, R.P. BRENT, R.S. BURACHIK, J.-A. H. OSBORN, B. SIMS & Q.J. ZHU (EDS) FROM ANALYSIS TO VISUALIZATION. JONATHAN BORWEIN COMMEMORATIVE CONFERENCE 2017, SPRINGER PROCEEDINGS IN MATHEMATICS AND STATISTICS 313, 423-439.
- TANAKA, K. (2017) TIME SERIES ANALYSIS: NONSTATIONARY AND NONINVERTIBLE DISTRIBUTION THEORY, SECOND EDITION. WILEY.
- TIAN, Y. (2015) CONGRUENT NUMBER PROBLEM AND ARITHMETIC OF ELLIPTIC CURVES. IN S.Y. CHENG, L. JI, Y.-S. POON, J. XIAO, L. YANG & S.-T. YAU (EDS.) INTRODUCTION TO MODERN MATHEMATICS: ADVANCED LECTURES IN MATHEMATICS 33, PP. 409-423, HIGHER EDUCATION PRESS.
- VAN GARDEREN, K.J. (1999) EXACT GEOMETRY OF FIRST ORDER AUTOREGRESSIVE MODELS. JOURNAL OF TIME SERIES ANALYSIS 20(1), 1-21.
- VAN GARDEREN, K.J. (2000) AN ALTERNATIVE COMPARISON OF CLASSICAL TESTS: ASSESSING THE EFFECTS OF CURVATURE. IN M. SALMON & P. MARRIOTT (EDS) APPLICATIONS OF DIFFERENTIAL GEOMETRY TO ECONOMETRICS, PP. 230-280. CAMBRIDGE UNIVERSITY PRESS.
- VOSKRESENSKAYA, G.V. (2012) ARITHMETIC PROPERTIES OF SHIMURA SUMS RELATED TO SEVERAL MODULAR FORMS. JOURNAL OF MATHEMATICAL SCIENCES 182(4), 444-455.
- WALDSCHMIDT, M. (2008) ELLIPTIC FUNCTIONS AND TRANSCENDENCE. IN K. ALLADI (ED.), SURVEYS IN NUMBER THEORY, PP. 148-188. DEVELOPMENTS IN MATHEMATICS 17, SPRINGER.
- WAN, J.G. (2012) MOMENTS OF PRODUCTS OF ELLIPTIC INTEGRALS. ADVANCES IN APPLIED MATHEMATICS 48, 121-141.
- WAN, J.G. (2013) RANDOM WALKS, ELLIPTIC INTEGRALS AND RELATED CONSTANTS. PH.D. THESIS, UNIVERSITY OF NEWCASTLE, NEW SOUTH WALES (AUSTRALIA). AVAILABLE AT: HTTP://DOCSERVER. CARMA.NEWCASTLE.EDU.AU/1487/1/JWTHESIS_REV.PDF.
- WAN, J.G. AND ZUCKER, I.J. (2016) INTEGRALS OF K AND E FROM LATTICE SUMS. THE RAMANUJAN JOURNAL 40(2), 257-278.
- WATSON, G.N. (1918) THE INTEGRAL FORMALA FOR GENERALIZED LEGENDRE FUNCTIONS. PROCEEDINGS OF THE LONDON MATHEMATICAL SOCIETY 17, 241-246.
- WHITE, J.S. (1961) ASYMPTOTIC EXPANSIONS FOR THE MEAN AND VARIANCE OF THE SERIAL CORRELATION COEFFICIENT. *BIOMETRIKA* 48, 85-94.
- ZAGIER, D.B. (2018) ARITHMETIC AND TOPOLOGY OF DIFFERENTIAL EQUATIONS. IN MEHRMANN, V. AND SKUTELLA, M. (EDS) PROCEEDINGS OF THE SEVENTH EUROPEAN CONGRESS OF MATHEMATICS (BERLIN, JULY 18-22, 2016), PP. 717-776. EUROPEAN MATHEMATICAL SOCIETY, BERLIN.

- ZHOU, Y. (2019A) ON BORWEIN'S CONJECTURES FOR PLANAR UNIFORM RANDOM WALKS. JOURNAL OF THE AUSTRALIAN MATHEMATICAL SOCIETY. 107(3), 392-411.
- ZHOU, Y. (2019B) SOME ALGEBRAIC AND ARITHMETIC PROPERTIES OF FEYNMAN DIAGRAMS. IN J. BLÜMLEIN, C. SCHNEIDER & P. PAULE (EDS), ELLIPTIC INTEGRALS, ELLIPTIC FUNCTIONS AND MODULAR FUNCTIONS IN QUANTUM FIELD THEORY, PP. 485-509. TEXTS AND MONOGRAPHS IN SYMBOLIC COMPUTATION 18, SPRINGER.
- ZUDILIN, W. (2013) PERIOD(D)NESS OF L-VALUES. IN J.M. BORWEIN, I. SHPARLINSKI & W. ZUDILIN (EDS.), NUMBER THEORY AND RELATED FIELDS: IN MEMORY OF ALF VAN DER POORTEN, SPRINGER PROCEEDINGS IN MATHEMATICS & STATISTICS 43, PP. 381-395.
- ZUDILIN, W. (2018) A HYPERGEOMETRIC VERSION OF THE MODULARITY OF RIGID CALABI-YAU MANIFOLDS. SPECIAL ISSUE ON MODULAR FORMS AND STRING THEORY IN HONOR OF NORIKO YUI. SYMMETRY, INTEGRABILITY AND GEOMETRY: METHODS AND APPLICATIONS (SIGMA) 14, 086 (14 PAGES) ARXIV: 1805.00544v2[MATH.NT].

Publisher's Note. Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Appendix: Proofs

We shall use the following decomposition derived by Bailey (1935, p. 29) of a

VWP non-terminating $_7F_6(1)$ series into two balanced $_4F_3(1)$ series under permutations of the parameters such that poles of the gamma function can be avoided and individual series converge.

$$\begin{split} W(a;c,d,e,f,g) &\coloneqq_{7} F_{6} \begin{bmatrix} a, 1+\frac{1}{2}a, c, d, e, f, g\\ \frac{1}{2}a, 1+a-c, 1+a-d, 1+a-e, 1+a-f, 1+a-g \end{bmatrix} \\ &= \frac{\Gamma(1+a-e)\Gamma(1+a-f)\Gamma(1+a-g)\Gamma(1+a-g-f-g)}{\Gamma(1+a)\Gamma(1+a-f-g)\Gamma(1+a-g-e)\Gamma(1+a-e-f)} \\ &\times_{4} F_{3} \begin{bmatrix} 1+a-c, d, f, g\\ 1+a-c, 1+a-d, e+f+g-a \end{bmatrix} \end{bmatrix} \\ &+ \frac{\Gamma(1+a-c)\Gamma(1+a-d)\Gamma(1+a-e)\Gamma(1+a-f)\Gamma(1+a-g)}{\Gamma(1+a)\Gamma(1+a-c-d)\Gamma(e)\Gamma(f)\Gamma(g)} \\ &\times \frac{\Gamma(e+f+g-a-1)\Gamma(2+2a-c-d-e-f-g)}{\Gamma(2+2a-c-e-f-g)\Gamma(2+2a-d-e-f-g)} \\ &\times_{4} F_{3} \begin{bmatrix} 2+2a-c-d-e-f-g, 1+a-f-g, 1+a-g-e, 1+a-e-f \\ 2+a-e-f-g, 2+2a-c-e-f-g, 2+2a-d-e-f-g \end{bmatrix} \\ &\times_{4} F_{3} \begin{bmatrix} 2+2a-c-d-e-f-g, 1+a-f-g, 1+a-g-e, 1+a-e-f \\ 2+a-e-f-g, 2+2a-c-e-f-g, 2+2a-d-e-f-g \end{bmatrix} \\ &\operatorname{Re}(2+2a-c-d-e-f-g) > 0. \end{split}$$

We shall also use a two-term relation he derived between VWP $_7F_6(1)$ series (*ibid.*, p. 62):

$$W(a; c, d, e, f, g) = \frac{\Gamma(1 + a - c)\Gamma(1 + a - d)\Gamma(2 + 2a - e - f - g)\Gamma(2 + 2a - c - d - e - f - g)}{\Gamma(1 + a)\Gamma(1 + a - c - d)\Gamma(2 + 2a - c - e - f - g)\Gamma(2 + 2a - d - e - f - g)} \times W(1 + 2a - e - f - g; c, d, 1 + a - f - g, 1 + a - e - g, 1 + a - e - f)}$$
(A.2)

PROOF OF THEOREM 1.

Wan (2013, p.115) showed that

$$\int_{0}^{1} K(k)^{2} dk = \int_{0}^{1} \frac{(1+k)K(k)K'(k)}{4\sqrt{k}} dk = \frac{1}{4} \int_{0}^{1} k^{-1/2} K(k)K'(k) dx + \frac{1}{4} \int_{0}^{1} k^{1/2} K(k)K'(k) dx. \quad (A.3)$$

Expression (6.65) in Wan (2013), which relates the left-hand-side to two balanced $_4F_3(1)$ series, follows from applying the Mellin transform given by Wan (2012, Proposition 2):

$$\int_{0}^{1} k^{n} K(k) K'(k) dx = \frac{\pi^{2}}{8} \frac{\Gamma\left[\frac{1}{2}(n+1)\right]^{2}}{\Gamma\left[\frac{1}{2}(n+2)\right]^{2}} {}_{4}F_{3} \left[\begin{array}{c} \frac{1}{2}, \frac{1}{2}, \frac{n+1}{2}, \frac{n+1}{2} \\ 1, \frac{n+2}{2}, \frac{n+2}{2} \end{array}; 1 \right].$$
(A.4)

Evaluating the ratio of gamma functions in terms of powers of π and L, and reciprocals gives

$$\int_{0}^{1} k^{-1/2} K(k) K'(k) dx = \frac{\pi L^{2}}{2} {}_{4}F_{3} \left[\begin{array}{c} \frac{1}{4}, \frac{1}{4}, \frac{1}{2}, \frac{1}{2} \\ \frac{3}{4}, \frac{3}{4}, \frac{3}{4}, \end{array}; 1 \right],$$
(A.5)

$$\int_{0}^{1} k^{1/2} K(k) K'(k) dx = \frac{\pi^{3}}{2L^{2}} {}_{4}F_{3} \begin{bmatrix} \frac{1}{2}, \frac{1}{2}, \frac{3}{4}, \frac{3}{4}, \\ 1, \frac{5}{4}, \frac{5}{4}; 1 \end{bmatrix}.$$
(A.6)

Expressions (3.9) and (3.11) then follow from noting that

$$F_{3}\begin{bmatrix}\frac{1}{4},\frac{1}{4},\frac{1}{2},\frac{1}{2}\\\frac{3}{4},\frac{3}{4},1\end{bmatrix} = {}_{7}F_{6}\begin{bmatrix}\frac{1}{4},\frac{9}{8},\frac{1}{8},\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{5}{8}\\\frac{1}{9},\frac{9}{8},\frac{1}{8},\frac{3}{4},\frac{3}{2},\frac{5}{8}\\\frac{1}{8},\frac{9}{8},1,\frac{3}{4},\frac{3}{4},\frac{5}{8}\end{bmatrix} = W\left(\frac{1}{4};\frac{1}{8},\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{5}{8}\right),$$
(A.7)

$${}_{4}F_{3}\begin{bmatrix}\frac{1}{2},\frac{1}{2},\frac{3}{4},\frac{3}{4},\frac{3}{4}\\1,\frac{5}{4},\frac{5}{4},\frac{5}{4}\end{bmatrix} = {}_{7}F_{6}\begin{bmatrix}\frac{3}{4},\frac{11}{8},\frac{3}{8},\frac{1}{2},\frac{1}{2},\frac{3}{4},\frac{7}{8}\\\frac{1}{8},\frac{11}{8},\frac{5}{4},\frac{5}{4},\frac{1}{8},\frac{7}{8},\frac{1}{8}\end{bmatrix} = W\left(\frac{3}{4};\frac{3}{8},\frac{1}{2},\frac{1}{2},\frac{3}{4},\frac{7}{8}\right), \quad (A.8)$$

and $\int_0^1 K(k)^2 dk = \frac{\pi^4}{32} W(\frac{1}{2}; \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2})$, upon multiplication by $2/\pi$. Thus (6.65) of Wan (2013), expressed in $_4F_3(1)$ series, is more fundamentally the three-term relation in VWP $_7F_6(1)$ series given by (3.9). Expressions (3.10) and (3.12) follow from consequences of the hitherto unnoticed fact that the $_4F_3(1)$ series in (3.5) is also a VWP $_7F_6(1)$ series:

$${}_{4}F_{3}\left[\frac{\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}}{1,1,1};1\right] = {}_{7}F_{6}\left[\frac{\frac{1}{2},\frac{5}{4},\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{3}{4};1\right]$$
$$= W\left(\frac{1}{2};\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{3}{4}\right) = W\left(\frac{1}{2};\frac{3}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{4}\right)$$
$$= \frac{2L^{2}}{\pi^{2}}{}_{4}F_{3}\left[\frac{\frac{1}{4},\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2}}{\frac{3}{4},\frac{3}{4},\frac{1}{1}};1\right] - \frac{2}{L^{2}}{}_{4}F_{3}\left[\frac{\frac{1}{2},\frac{1}{2},\frac{3}{4},\frac{3}{4}}{1,\frac{5}{4},\frac{5}{4},\frac{1}{1}};1\right] by (A.1).$$

(3.10) and (3.12) follow from applying (3.5), (A.7) and (A.8) and multiplying by $\pi^2/2$.

PROOF OF THEOREM 2.

Wan (2013, p.115) also showed that

$$\int_{0}^{1} K(k)^{2} dk = \int_{0}^{1} \frac{(1-k)K'(k)^{2}}{8\sqrt{k}} dk = \frac{1}{8} \int_{0}^{1} k^{-1/2} K'(k)^{2} dk - \frac{1}{8} \int_{0}^{1} k^{1/2} K'(k)^{2} dk.$$
(A.9)

Now we use the Mellin transform derived by Wan (2012, Proposition 1):

$$\begin{split} \int_{0}^{1} k^{n} K'(k)^{2} dx &= \frac{2^{4n} (n+1)}{16} \frac{\Gamma\left[\frac{1}{2} (n+1)\right]^{8}}{\Gamma(n+1)^{4}} {}^{7} F_{6} \left[\begin{array}{c} \frac{n+1}{2}, & \frac{n+5}{4}, & \frac{n+1}{2}, & \frac{1}{2}, & \frac{1}{2}, & \frac{1}{2}, & \frac{1}{2}, \\ \frac{n+1}{4}, & 1, & \frac{n+2}{2}, & \frac{n+2}{2}, & \frac{n+2}{2}, \\ \frac{n+2}{2}, & \frac{n+2}{2}, & \frac{n+2}{2}, & \frac{n+2}{2}, \\ \frac{n+2}{2}, & \frac{n+2}{2},$$

Substituting for n and evaluating the ratio of gamma functions in terms of powers of π , L and their reciprocals gives

$$\int_{0}^{1} k^{-1/2} K'(k)^{2} dx = \frac{L^{4}}{2} W\left(\frac{1}{4}; \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\right),$$
(A.11)

$$\int_{0}^{1} k^{1/2} K'(k)^{2} dx = \frac{3}{2} \left(\frac{\pi}{L}\right)^{4} W\left(\frac{3}{4}; \frac{3}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\right).$$
(A.12)

Expression (3.13) follows from applying (A.9), (A.11) and (A.12) upon multiplication by 16, noting that $\int_0^1 K(k)^2 dk = (\pi^4/32) \times W(\frac{1}{2}; \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2})$. Applying the decomposition (A.1) to the Wexpressions in (A.11) and (A.12) gives

$$W\left(\frac{1}{4};\frac{1}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}\right) = -\frac{\pi^2}{L^4} {}_4F_3\left[\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}\right] + \frac{4}{L} {}_4F_3\left[\frac{1}{4},\frac{$$

$$W\left(\frac{3}{4};\frac{3}{4},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2}\right) = \frac{1}{3}\frac{L^4}{\pi^2} {}_4F_3\left[\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2},\frac{1}{2};1\right] - \frac{2}{3\pi}L_4F_3\left[\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4};1\right].$$
 (A.14)

(3.15) now follows from applying (A.9), (A.11) and (A.12) upon multiplication by 8.

Expressions (6.41) and (7.25) by Wan (2013) give

$$\frac{\pi^3}{4} {}_4F_3 \begin{bmatrix} \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2} \\ 1, 1, 1 \end{bmatrix} = \int_0^1 \frac{(1+k)K(k)^2}{\sqrt{k}} \, dk = \int_0^1 \frac{(1+k)K'(k)^2}{4\sqrt{k}} \, dk.$$
(A.15)

Expression (3.14) then follows from $W(\frac{1}{2}; \frac{1}{4}, \frac{1}{2}, \frac{1}{2}, \frac{1}{2}, \frac{3}{4}) = {}_{4}F_{3}\left[\begin{array}{c} \frac{1}{2}, \frac{1}{2$

PROOF OF THEOREM 3.

Part (a) follows by applying (A.2) to $W(1; \frac{1}{2}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4})$. For (b) note that

$$W\left(1;\frac{1}{2},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4}\right) = {}_{7}F_{6}\left[\frac{1,\frac{3}{2},\frac{1}{2},\frac{3}{4},\frac{$$

Now apply (A.1) to $W(1; \frac{1}{2}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4}, \frac{3}{4})$ to get

$${}_{5}F_{4}\left[\frac{\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},1}{\frac{5}{4},\frac{5}{4},\frac{5}{4},\frac{5}{4},\frac{1}{2}};1\right] = -\frac{L}{4}{}_{4}F_{3}\left[\frac{\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4}}{\frac{5}{4},\frac{5}{4},\frac{5}{4},\frac{3}{2},\frac{3}{2}}\right] + \frac{L^{4}}{8\pi}{}_{4}F_{3}\left[\frac{\frac{1}{2},\frac{1$$

on applying (A.16). (3.19) follows upon multiplication by $8\pi^3/L^4$. Finally, (3.20) follows from subtracting (3.19) from (3.15), using (3.16) and then dividing both sides by π^3 .

PROOF OF LEMMA 4.

Expression (5.6) is (M.11a) in McCrorie (2020b) where it is one of a taxonomically constructed set of $_{3}F_{2}(1)$ series allied to M. It can be proved directly using (7.4.4.1) and (7.4.4.5) from Prudnikov, Brychkov and Marichev (1986). Specifically, the former is

$${}_{3}F_{2}\left[\begin{array}{c}a,b,c\\d,e\end{array};1\right] = \frac{\Gamma(d)\Gamma(d+e-a-b-c)}{\Gamma(d+e-a-b)\Gamma(d-c)}{}_{3}F_{2}\left[\begin{array}{c}e-a,e-b,c\\d+e-a-b,e\end{array};1\right].$$
 (A.17)

$$(\text{Re}(d + e - a - b - c) > 0 \text{ and } \text{Re}(d - c) > 0)$$

and the latter is

$${}_{3}F_{2}\left[\begin{array}{c}a,b,c\\d,e\end{array};1\right] = \frac{\Gamma(1+a-d)\Gamma(1+b-d)\Gamma(1+c-d)\Gamma(d)\Gamma(e)}{\Gamma(a)\Gamma(b)\Gamma(c)\Gamma(1+e-d)\Gamma(2-d)}$$
(A.18)
$${}_{3}F_{2}\left[\begin{array}{c}1+a-d,1+b-d,1+c-d\\1+e-d,2-d\end{array};1\right] + \frac{\Gamma(1+a-d)\Gamma(1+c-d)}{\Gamma(1-d)\Gamma(1+a+c-d)}$$
$${}_{3}F_{2}\left[\begin{array}{c}a,c,e-b\\1+a+c-d,e\end{cases};1\right].$$
(Re(d+e-a-b-c)>0 and Re(1+b-d)>0)

In one application of (A.18) below, the ${}_{3}F_{2}(1)$ series reduces to a well-defined ${}_{2}F_{1}(1)$ series, which is summable using the Gauss Summation Theorem (e.g. Bailey (1935), equation 1.3.1):

$${}_{2}F_{1}(a,b;c;1) = \frac{\Gamma(c)\Gamma(c-a-b)}{\Gamma(c-a)\Gamma(c-b)} (\operatorname{Re}(c-a-b) > 0).$$
(A.19)

By (A.17),

$$M = {}_{3}F_{2} \begin{bmatrix} \frac{1}{2}, 1, 1\\ \frac{5}{4}, \frac{3}{2}; 1 \end{bmatrix} = \frac{L^{2}}{\pi} {}_{3}F_{2} \begin{bmatrix} \frac{1}{2}, \frac{1}{2}, \frac{1}{2}\\ \frac{3}{4}, \frac{3}{2}; 1 \end{bmatrix}.$$
(A.20)

By (A.18),

$${}_{3}F_{2}\begin{bmatrix}\frac{1}{2},\frac{1}{2},\frac{1}{2}\\\frac{3}{4},\frac{3}{2};1\end{bmatrix} = \frac{2\sqrt{2}}{3}\frac{\pi}{L^{2}}{}_{3}F_{2}\begin{bmatrix}\frac{3}{4},\frac{3}{4},\frac{3}{4}\\\frac{5}{4},\frac{7}{4};1\end{bmatrix} = \frac{\pi}{L^{2}}{}_{3}F_{2}\begin{bmatrix}\frac{1}{2},\frac{1}{2},1\\\frac{5}{4},\frac{3}{2};1\end{bmatrix}.$$
(A.21)

We now apply (A.17) to the first ${}_{3}F_{2}(1)$ series on the right-hand-side of (A.21):

$${}_{3}F_{2}\begin{bmatrix}\frac{3}{4},\frac{3}{4},\frac{3}{4}\\\frac{5}{4},\frac{7}{4};1\\\frac{5}{4},\frac{7}{4}\end{bmatrix} = \frac{1}{\sqrt{2}}{}_{3}F_{2}\begin{bmatrix}\frac{3}{4},1,1\\\frac{3}{2},\frac{7}{4};1\\\frac{3}{2},\frac{7}{4}\end{bmatrix}$$
(A.22)

We then apply (A.18) to the second:

$${}_{3}F_{2}\left[\frac{\frac{1}{2},\frac{1}{2},1}{\frac{5}{4},\frac{3}{2}};1\right] = \sqrt{2}L_{2}F_{1}\left[\frac{\frac{1}{4},\frac{1}{4}}{\frac{5}{4}};1\right] - {}_{3}F_{2}\left[\frac{\frac{1}{2},1,1}{\frac{5}{4},\frac{3}{2}};1\right], \quad (A.23)$$

where $_{2}F_{1}\begin{bmatrix}\frac{1}{4},\frac{1}{4}\\\frac{5}{4};1\end{bmatrix} = \frac{\pi}{2\sqrt{2}}$ (by A.19). Hence, by (A.20) – (A.23), we obtain (5.13):

$${}_{3}F_{2}\left[\begin{array}{c} \frac{\overline{4}}{4}, 1, 1\\ \frac{3}{2}, \frac{7}{4}; 1\\ \frac{3}{2}, \frac{7}{4} \end{array}\right] = 3\left(M - \frac{\pi}{4}L\right)$$

PROOF OF THEOREM 5.

The advantage of using (5.13) is that it has a VWP $_7F_6(1)$ series representation that immediately reduces to the $_3F_2(1)$ series by the cancellation of numerator and denominator parameters:

$${}_{3}F_{2}\left[\frac{\frac{3}{4},1,1}{\frac{3}{2},\frac{7}{4}};1\right] = {}_{7}F_{6}\left[\frac{\frac{3}{2},\frac{7}{4},\frac{3}{4},\frac{3}{4},\frac{3}{4},1,1,\frac{5}{4}}{\frac{3}{4},\frac{7}{4},\frac{7}{4},\frac{3}{2},\frac{3}{2},\frac{3}{2},\frac{5}{4}};1\right] =:W\left(\frac{3}{2};\frac{3}{4},\frac{3}{4},1,1,\frac{5}{4}\right).(A.24)$$

Additional VWP $_7F_6(1)$ series can be derived by permuting the entries in (A.24) and applying the two-term relation (A.2):

$$W\left(\frac{3}{2};\frac{3}{4},\frac{3}{4},1,1,\frac{5}{4}\right) = \frac{9}{16}\frac{\pi^2}{L}W\left(\frac{3}{4};\frac{3}{4},\frac{3}{4},\frac{1}{4},\frac{1}{4},\frac{1}{2}\right)$$

$$W\left(\frac{3}{2};\frac{3}{4},1,\frac{3}{4},1,\frac{5}{4}\right) = 2W\left(1;\frac{3}{4},1,\frac{1}{4},\frac{1}{2},\frac{3}{4}\right)$$

$$W\left(\frac{3}{2};1,1,\frac{3}{4},\frac{3}{4},\frac{5}{4}\right) = \frac{5}{3}W\left(\frac{5}{4};1,1,\frac{1}{2},\frac{1}{2},1\right)$$

$$W\left(\frac{3}{2};\frac{3}{4},\frac{5}{4},1,1,\frac{3}{4}\right) = \frac{5L}{3}W\left(\frac{5}{4};\frac{5}{4},\frac{5}{4},\frac{3}{4},\frac{1}{2},\frac{3}{4}\right).$$
(A.25)

Finally, applying (A.2) to (A.25) with its entries appropriately permuted gives

$$W\left(1;\frac{1}{2},\frac{1}{4},1,\frac{3}{4},\frac{3}{4}\right) = \frac{3}{4}\left(\frac{\pi}{L}\right)^2 W\left(\frac{1}{2};\frac{1}{2},\frac{1}{4},\frac{1}{2},\frac{1}{4},\frac{1}{4},\frac{1}{4}\right).$$

The six expressions in the statement of theorem are obtained by dividing the right-hand expressions by the factor 3 that appears in (5.13), reordering the entries of each $_7F_6(1)$ series in ascending order, and putting the series following the canonical series in ascending order in the parameter a.

J. RODERICK MCCRORIE

UNIVERSITY OF ST ANDREWS, SCOTLAND, UK

Paper received: 3 July 2019. Paper accepted: 7 May 2020.